ASSESSING STAFF READINESS FOR SIMULATION IN A HEALTH SCIENCES INSTITUTION

Tolulope Ayomipo Awogbemila

A research report submitted to the Faculty of Health Sciences, University of the

Witwatersrand, Johannesburg, in partial fulfillment of the requirements for the degree of

Master of Science in Nursing

Johannesburg, 2018

DECLARATION

I, Tolulope Ayomipo Awogbemila declare that this research report (Human Research Ethics

Clearance number M170238) is my own work. It is being submitted for the degree of Master of Science in Nursing in the University of the Witwatersrand, Johannesburg. It has not been submitted before for any degree or examination at this or any other University.

.....

Signed at Johannesburg

On the, 2018

DEDICATION

I dedicate this work to the Almighty God for the grace given to study and successfully complete the research work.

PRESENTATIONS ARISING FROM THIS STUDY

Awogbemila, T.A., Thurling, C.H., Armstrong, S.J. & George, A. (2017). Assessing staff readiness for simulation in a health sciences institution. In Sigma Theta Tau International Chi Xi-At Large Chapter first Biennial conference, Swaziland.

Awogbemila, T.A., Thurling, C.H., Armstrong, S.J. & George, A. (2017). Assessing staff readiness for simulation in a health sciences institution. In School of Therapeutic Sciences research day at the University of the Witwatersrand, South Africa.

ABSTRACT

Background: Simulation has been considered a possible solution to the recorded imbalance in the number of students, clinical instructors and clinical placements. The perceived simulation benefits have led to an increase in investment by health sciences institutions through the purchase of equipment and set up of a simulation laboratory, but with a lack of proper planning and structure for its integration into the curriculum. Frotjold (2015) indicated that poor staff preparation and planning can inhibit the adoption of simulation thereby limiting its utilization.

Purpose: To examine the readiness of lecturers from the five departments within the school of therapeutic health sciences at a university in South Africa to adapting simulation-based education and identify factors preventing or promoting the successful use of the new methodology.

Methodology: A quantitative cross-sectional descriptive design was adopted utilizing a validated Simulation Culture Organizational Readiness Survey (SCORS) 24 item, 5-point likert scale. Sections of the survey addressed: Defined need and support for change; readiness for culture change; time, personnel and resource readiness; and sustainability practices to embed culture. A total population sampling method was adopted, and lecturers identified were contacted electronically using the Redcap software.

Result: The results of this study showed that lecturers in The School were "*somewhat ready*" for simulation (107.5). This was due to a lack of strategic vision (2.78), the lack of resources (2.3), and insufficient staff education (2.52). The fact that innovation and experiential learning was central to the institution's mission and philosophy (4.02), the technological proficiency of lecturers (3.56) and positive attitudes (3.5) promoted the use of simulation.

Conclusion/Recommendation: Evaluating staff readiness and providing training to empower staff is recommended for institutions initiating simulation to achieve successful student outcomes with simulation use.

ACKNOWLEDGEMENTS

My gratitude and thanks go to the following people who made this study possible:

To my supervisors, Dr. Hilary Thurling, Dr. Susan J. Armstrong and Dr. Ann George for their support, availability and motivation throughout this study. It was great learning under your guidance and support.

To the Head (Prof. Lize Maree) and all the lecturers of the Department of Nursing Education, Wits University-you have been supportive and motivating.

To my husband (Bolanle O. Eweoya) and entire family: thank you for all the support in all aspects and for being there for me.

To Christmal Kpodo: thank you for all the assistance and valuable insight.

To Senyo Dzokoto: thank you for helping with a part of the data analysis.

Participants: This work would not be possible without your agreement to participate.

To my examiners for all their input and comments that made this work complete.

To the National Research Foundation for funding this research work.

God bless you all.

CONTENTS	PAGE
DECLARATION	ii
DEDICATION	iii
PRESENTATIONS ARISING FROM THIS STUDY	iv
ABSTRACT	v
ACKNOWLEDGEMENTS	vi
LIST OF ABBREVIATIONS	v

CHAPTER ONE: OVERVIEW OF THE STUDY

1.1 Introduction	1
1.2 Background of the study	1
1.3 Problem statement	2
1.4 Research questions	3
1.5 Research purpose	4
1.6 Research objectives	4
1.7 Significance of the study	4
1.8 Research setting	4
1.9 Population and sampling	5

1.10 Overview of research design and method	5
1.11 Definition of terms	5
1.12 Conclusion	6

CHAPTER TWO: LITERATURE REVIEW

2.1 Introduction	7
2.2 The concept of simulation	7
2.2.1 Classification of simulation used in health professions education	8
2.3 The call for simulation in health professions education	8
2.4 The use of simulation in health sciences professions and its benefits	9
2.4.1 Simulation and occupational therapy education	9
2.4.2 Simulation and physiotherapy education	10
2.4.3 Simulation and pharmacy and pharmacology education	11
2.4.4 Simulation and nursing education	11
2.4.5 Simulation and sports medicine	12
2.5 Challenges of implementing simulation	
2.6 Readiness factors for simulation adoption	14
2.6.1 Science	14
2.6.2 Staff	14
2.6.3 Space and supplies	15
2.6.4 Support	15
2.6.5 Success and sustainability	16

2.7 Conceptual frameworks for readiness	
2.7.1 Structural and external factors	18
2.7.2 Staff attributes	19
2.7.3 Other psychological factors	19
2.8 Conclusion	20
CHAPTER THREE: RESEARCH METHODOLOGY	
3.1 Introduction	21
3.2 Research design	21
3.3 Research setting	21
3.4 Population and sample	
3.4.1 Inclusion criteria	22
3.4.2 Exclusion criteria	22
3.4.3 Sampling technique	22
3.4.4 Estimated sample size	23
3.5 Data collection procedure	
3.5.1 Instrument	23
3.5.2 Validity and reliability	24
3.6 Data analysis	25
3.7 Ethical considerations	25

	3.7.1 Permission to conduct the study	26
	3.7.2 Informed consent	26
	3.7.3 Confidentiality and anonymity	27
	3.7.4 Non-maleficence	27
	3.7.5 Justice	27
	3.7.6 Respect	28
3.8 Co	onclusion	28

CHAPTER FOUR: DATA ANALYSIS AND RESULTS

4.1 Introduction	29
4.2 Response rate	30
4.3 Demographic information	30
4.3.1 Age distribution	30
4.3.2 Gender	31
4.3.3 Department	31
4.3.4 Employment status	32
4.3.5 Years of service of respondents in their departments	32
4.4 Readiness of lecturers within The School	33
4.4.1 The readiness of lecturers in the five departments within The School	33

4.5 Factors preventing or promoting the readiness of lecturers to use simulation within	
the school	34
4.5.1 Section A: defined need and support for change within The School	34
4.5.2 Section B: readiness for culture change within The School	35
4.5.3 Section C: time, personnel and resource readiness	36
4.5.4 Section D: sustainability practices to embed culture within The School	37
4.5.5 Readiness of lecturers within The School by sections	38
4.6 Factors preventing or promoting the readiness of lecturers to use simulation	
within each department of The School	38
4.6.1 Section A: defined need and support for change by department	38
4.6.2 Section B: readiness for culture change by department	39
4.6.3 Section C: time, personnel and resource readiness by department	40
4.6.4 Section D: sustainability practices to embed culture by department	41
4.6.5 Sectional readiness of lecturers by department	42
4.6.6 The difference in the readiness for simulation in the five departments	42
4.7 Factors preventing or promoting the readiness of lecturers to use simulation by	
age group	43
4.7.1 Sectional readiness by age group	45
4.7.2 The difference in the readiness for simulation by age group	46
4.8 Factors preventing or promoting the readiness of lecturers to use simulation by	
employment status	46
4.8.1 Sectional readiness by employment status	47
4.8.2 The difference in the readiness for simulation by employment status	48

4.9 Sectional readiness by years of experience	
4.9.1 The difference in the readiness for simulation by years of experience	48
4.10 Conclusion	49
CHAPTER FIVE: DISCUSSION OF FINDINGS, CONCLUSION, AND	
RECOMMENDATIONS	
5.1 Introduction	50
5.2 Discussion of findings	
5.2.1 Readiness of The School to integrate SBE	50

5.2.2 Antecedents of readiness supporting SBE51

5.2.2.1 Structural and external factors	51	
	5.2.2.2 Staff attributes	51

- 5.2.2.3 Psychological factors525.2.3 Antecedents to readiness hindering SBE52
- 5.2.3.1 Structural and external factors525.2.3.2 Staff attributes54
- 5.2.3.3 Psychological factors54

5.3 Limitation	55
5.4 Conclusion	55

5.5 Recommendations	57

REFERENCES

60

APPENDICES

Appendix A: The university postgraduate assessor's committee approval	69
Appendix B: Ethics clearance from the university's human research ethics committee	
(medical)	70
Appendix C: Permission to use the simulation culture organizational readiness	
survey (SCORS) instrument	71
Appendix D: Participant's information document and consent form	73
Appendix E: Adapted survey (SCORS) instrument	74
Appendix F: The original SCORS instrument	80

69

LIST OF FIGURES

CHAPTER TWO

Figure 2.1 Conceptual frameworks for readiness	18
Figure 4.1 Departmental readiness for simulation	33
LIST OF TABLES	
CHAPTER FOUR	
Table 4.1 Respondents to the distributed survey	30
Table 4.2 Age group of participants	31
Table 4.3 Gender of respondents	31
Table 4.4 Number of respondents from each of the five departments	32
Table 4.5 Employment status of respondents	32
Table 4.6 Years of service of respondents	33
Table 4.7 Defined need and support for change within the School of Therapeutic	
Sciences	35
Table 4.8 Readiness for culture change within the School of Therapeutic Sciences	36
Table 4.9 Time, personnel and resource readiness within the School of Therapeutic	
Sciences	37
Table 4.10 Sustainability practices to embed culture within the School of Therapeutic	

$\mathbf{\alpha}$	•		
•	0.1	0.10	000
· ``	(' I	еп	1.60
	\mathbf{v}	\mathbf{v}	

Table 4.11	An overview	of readiness	of lecturers	within	the School of	Therapeutic
-------------------	-------------	--------------	--------------	--------	---------------	-------------

Sciences by sections	38
Table 4.12 Defined need and support for change by department	39
Table 4.13 Readiness for culture change by department	40
Table 4.14 Time, personnel and resource readiness by department	41
Table 4.15 Sustainability practices to embed culture by department	42
Table 4.16 Sectional readiness of lecturers in specific departments	42
Table 4.17 Difference in the readiness for simulation in the five departments	43
Table 4.18 Readiness for culture change by age group	44
Table 4.19 Time, personnel and resource readiness by age group	45
Table 4.20 Sectional readiness by age group	46
Table 4.21 Readiness for culture change by employment status	47
Table 4.22 Section readiness by employment status	48
Table 4.23 Sectional readiness by years of experience	48

LIST OF ABBREVIATIONS

CESSM- Center for exercise science and sports medicine HFS- High Fidelity Simulation LFS- Low Fidelity Simulation MFS – Medium Fidelity Simulation NES – Nursing Educator Stakeholders NUR- Nursing OT- Occupational Therapy PHY- Physiotherapy PHA- Pharmacy and Pharmacology SBE- Simulation-Based Education SCORS - The Simulation Culture Organizational Readiness Survey The School- School of Therapeutic Sciences

CHAPTER ONE

OVERVIEW OF THE STUDY

1.1 Introduction

Chapter One provides an overview to this study by outlining the following: The background to the study, problem statement, research question, purpose of the study, research objectives, significance of the study, definition of terms as applied in the study, as well as a brief overview of the research design and methodology.

1.2 Background

Simulation is a teaching and learning methodology historically used in the military and aviation industries and currently adopted by the health sciences to mimic real life clinical experiences in the training of health professionals, to improve both patient care and student outcomes (Aebersold & Tschannen, 2013). Gudayu et al. (2015) expressed the opinion that a major aspect of simulation education is the ability to reproduce important aspects of a clinical situation and facilitate better understanding and management of these clinical situations in practice, with the opportunity to repeat tasks to consolidate learning and be competent through instructor feedback and debriefing. Simulation is being used by health sciences institutions, as a teaching tool, and a mechanism for skills assessment during objective structured clinical assessment/examination (OSCA/OSCE) (Galloway, 2009; Stunden et al., 2015).

Simulation-Based Education (SBE) proffers great benefits to student learning in an experiential, safe and constructive environment, helping health professional students to develop critical thinking skills and confidence for the workplace (McGrath et al., 2012). Research findings (Stunden et al., 2015; Ohtake & Erdley, 2013) have shown that the students' engagement in the simulation experience has improved their satisfaction, confidence and self-efficacy in performing skills required for clinical practice.

The benefits of SBE are not only student centered but assists health sciences institutions to provide solutions to issues that arise around or with clinical placements. These issues include: an imbalance of student-patient ratio, students from multiple educational institutions competing for

health facility placements as well as a lack of uniformity of skills learnt amongst students, as students are placed in different units and might not be given an opportunity to experience managing a patient with a particular condition (Rodriguez, 2013; McCabe, 2016). A compounding issue is the lack of sufficient supervisors and preceptors in the wards to supervise learners and coordinate learning activities (The Nursing Education Stakeholders (NES) Group, 2012).

With the development and growth of simulation in health sciences education and training, there has been an increase in investment in the purchase of equipment and setting up of simulation laboratories by nursing education institutions, but with a noticeable lack of proper planning and structure for simulation integration into the curriculum (Leighton & Foisy-Doll, 2016). A lack of planning is reflected in the limited number of staff to champion SBE and a limited number of staff who possess the knowledge and skills needed to use simulation as a teaching and assessment tool (Lazzara et al., 2014; Nehring et al., 2013). This situation may occur where institutions strive to reduce costs by employing a smaller workforce (Lazzara et al., 2014), which could contribute to a possible lack of readiness for simulation and a potential failure in implementing the simulation program (Shea et al., 2014). The lack of planning and readiness for simulation is anticipated to be worse in allied health faculties as simulation is newer for these departments than in nursing (Dennis et al., 2016).

Institutional readiness refers to the adoption of an organizational culture and policy, ensuring the availability of skilled personnel, structure, finance and support to prepare staff psychologically and behaviorally for a change (Society of Hospital Medicine, 2016; Weiner et al., 2009). As postulated by Shea et al. (2014:2), "the higher the organizational readiness, the more the staff is likely to initiate change, exert greater effort, exhibit greater persistence, and display more cooperative behavior, resulting in a more effective implementation of the proposed change. Equally, a low organizational readiness, results in members viewing the change as undesirable and resist planning for the change and engaging in the change process".

1.3 Problem statement

With the amount of investment made by health sciences institutions into simulation, Kenney (2014) claims that the effectiveness of simulation in allied healthcare education is often minimal

due to a lack of staff knowledge or technical know-how of the equipment, how to construct the scenarios, and how to evaluate the simulation. The equipment is being under-utilized, with medical institutions in general, often having the acquired simulator equipment remaining in unopened boxes and not being used for student teaching and learning (Leighton & Foisy-Doll, 2016). Research evidence (Frotjold, 2015) has also shown how a lack of a guiding philosophy and poor staff preparation can inhibit the adoption of simulation thereby limiting its utilization.

The School of Therapeutic Sciences (The School) of a University in South Africa established and equipped a simulation laboratory for use by the health sciences students and staff. The simulation laboratory has been predominantly used by the nursing department (NUR) and to a lesser degree by the physiotherapy department (PT), with simulation training being offered to the other departments; pharmacy and pharmacology (PHA), occupational therapy (OT) and center for exercise science and sports medicine (CESSM) within The School as simulation is a new teaching methodology in The School.

The school has been observed not to have conducted an assessment of readiness prior to the setup of the laboratory, which signals a possible challenge to integrate simulation. Miller & Bull (2013) have indicated that institutions that embark upon simulation through the purchase of equipment without adequate planning for simulation, often experience staff resistance, and the staff failing to use the equipment.

As emphasized by Miller & Bull (2013), innovations should be surveyed 'through the lens' of the academics to better understand adoptions or rebuffs of a new pedagogy. Miller & Bull (2013) view has informed the researcher's aim to assess the readiness for simulation adoption through the perception of lecturers within the five departments of the School of Therapeutic Sciences of the University. The School of Therapeutic Sciences will be referred to as "*the School*" for the remainder of this report.

1.4 Research question

• What is the extent of readiness of lecturers from the five departments within The School of Therapeutic Sciences at the selected University for the use of SBE?

1.5 Research purpose

• To investigate the readiness of lecturers from the five departments within The School to adopting simulation-based education and identify the factors preventing or promoting the successful use of the new methodology.

1.6 Research objectives

- To determine the extent of readiness of lecturers from the five departments to integrate simulation-based education in The School.
- To investigate the factors that could prevent or promote the full utilization of simulation in The School.

1.7 Significance of the study

It is important to examine the readiness of lecturers within the five departments of The School for the adoption of simulation, as it is the foundation to ensuring successful integration of simulation-based education into the curriculum of The School and facilitating optimum use of the newly set up simulation laboratory in The School. This could in-turn improve the successful outcomes and efficiency of simulation-based programs.

The results of this study stand to provide administrators and educators with insight into organizational readiness factors that either facilitate or challenge the successful utilization of clinical simulation. It will also show that the successful integration of simulation requires a shift in the existing teaching pedagogy from didactic to a more student-centered experiential type of teaching and learning within The School.

1.8 Research setting

The School is one of the seven schools in the Faculty of Health Sciences within the University. The School which comprises of the Departments of Nursing Education (NUR), Occupational Therapy (OT), Physiotherapy (PHY), Pharmacy and Pharmacology (PHA) and the Centre for Exercise Science and Sports Medicine (CESSM) was the chosen setting for the research.

1.9 Population and Sampling

The study population consisted of all full-time and part-time lecturers involved in the teaching of undergraduate students from the five departments within The School of the University. The lecturers had to be employed on a minimum of a 50% post. A total population sampling method was adopted which is "a type of purposive sampling technique where you choose to examine the entire population that have a particular set of characteristics" (Laerd, 2012:25). The researcher chose to study the entire population as the size of the population that had the set of characteristics of interest was very small and because the researcher wanted to establish the readiness of all the lecturers in the school of therapeutic health sciences to simulation.

1.10 Overview of research design and method

A quantitative cross-sectional descriptive design was used, utilizing a validated survey instrument to assess staff readiness for the adoption of simulation in the five departments within The School.

The survey was sent to the research participants electronically via the REDCap[®] software after permissions had been obtained from all authorities including obtaining participants' voluntary consent to participate in the study. The results were analyzed using descriptive statistics.

1.11 Definition of terms

- **Simulation**: In health professions education, simulation is described as a technique or activity that mimics a clinical setting or its activities, to facilitate practice of procedures, decision-making and critical thinking by participants using role play, video or simulators (Jeffries, 2005).
- **Role play**: entails acting out an event or situation which enables learners to react to the event as they would in a real-life scenario (Galloway, 2009).
- **Standardized patients**: are individuals, also referred to as actors, who play diverse roles such as replicating patient's state of health or family member's behavior to create a learning experience for the learners (Ker & Bradley, 2010).

- **Task trainers:** are parts of a patient simulator used in the teaching and learning of psychomotor skills. A typical example is the torso used in cardio-pulmonary resuscitation training (Ker & Bradley, 2010).
- **Human patient simulator**: is a full-sized patient simulator that can perform the action of eye movement, respiration, and circulation (Lateef, 2010).
- **Simulation-based education**: is referred to as any educational activity that uses simulation aides to imitate clinical situations and is aimed at ensuring learners develop knowledge, skills and attitudes in a safe environment (Al-Elq, 2010; Lateef, 2010).
- **Readiness**: The state of being fully prepared for something or the willingness to do something (Oxford dictionary, 2017). Readiness is the state of preparedness of individuals, systems, or institutions to meet a situation and carry out a planned sequence of actions. Readiness is based on meticulousness of the planning, adequacy and training of the staffs, and supply and reserve of support services or systems (Business dictionary, 2017).
- **Institutional readiness**: relates to whether an institution has the infrastructure and support needed to embark upon a new initiative. It covers not only the presence and availability of skills, finance and infrastructure, but also refers to the culture and policies in place within the organization (Society of Hospital Medicine, 2016).

1.12 Conclusion

In this Chapter, the background of this study, the problem statement, research question, purpose and objectives were explained. Definitions of concepts were provided, and the research design and method were briefly covered, to be explained in detail in Chapter Three.

The following chapters are to follow:

Chapter Two provides a literature review.

Chapter Three describes the research methodology used.

Chapter Four covers the presentation of the findings from the study.

Chapter Five gives a discussion of the findings, the conclusions and the recommendations arising from the study.

CHAPTER TWO

LITERATURE REVIEW

2.1 Introduction

This chapter presents the literature review surrounding simulation and its adoption in health sciences institutions. The readiness factors that could facilitate adoption of simulation and a conceptual framework to support the study are also discussed.

2.2 The concept of simulation

Simulation is a technique that creates a situation or environment to permit individuals to experience a depiction of a real event for practice, learning, evaluation, testing, or to gain an understanding of systems or human actions (Lopreiato et al., 2016).

Simulation's history stems from its use in the military, nuclear power, and aviation industries. Flight simulators are used in the training of pilots, disaster training is conveyed using simulation in nuclear power industries and the use of war games is adopted in military training (Galloway, 2009; Aebersold & Tschannen, 2013). The successful use of simulation in these professions contributed to its adoption in the education and training of health professionals (Galloway, 2009; Harder, 2010).

The education and training of health professionals using simulation enable learners to practice required skills in an environment that provides room for mistakes and professional growth without endangering patient safety (Galloway, 2009; Harder, 2010). Simulation also provides learners with the opportunity to repeat skills for better understanding and be competent through constructive instructor feedback and debriefing (Gudayu et al., 2015). To ensure a meaningful simulation experience, a pattern or plan should be designed by the facilitator to entail introductory learning by the participant before the simulation, followed by the implementation of the simulation, which is then followed by a debriefing session to permit reflection on activities (Aebersold & Tschannen, 2013).

2.2.1 Classification of simulation used in health professions education

Simulation is classified into three different categories; low, medium and high fidelity. The categories are termed according to their likeness to reality. **Low-fidelity simulation (LFS) manikins** are static in nature and deficient in terms of realism or situational milieu. Low-fidelity are referred to as "task trainers", such as a prosthetic arm for intravenous insertion or a prosthetic buttock used for intramuscular injections administration. Task trainers are used in teaching basic technical skills. Role-playing scenarios is also an example of low fidelity simulation. **Medium fidelity simulation (MFS)** portrays more realism with the ability of simulators to produce heart sounds and breath sounds but they are unable to talk, show chest or eye movement. **High fidelity simulation (HFS)** uses computerized human patient simulators and is operated by facilitators to make physiological changes and sounds (Evans, 2017; Shea et al., 2015; Harder, 2010 & AI-Elq, 2010). Standardized patients are also regarded as high-fidelity simulation as the standardized patient reacts to student actions verbally and in context (Ozelie et al., 2016).

Irrespective of the type of SBE or simulator used, proper planning, preparation and understanding of what is to be achieved in the learning experience, is required of educators and administrators to ensure a positive learning experience for the students (Larue et al., 2015; Sowerby, 2015; National League for Nursing, 2015).

2.3 The call for simulation in health professions education

Precipitating factors of learners' poor learning experience and skills acquisition in health sciences institutions, and dissatisfied employers of newly qualified health care practitioners have been linked with the existing nursing and nurse faculty shortage, large student numbers and an insufficient number of clinical sites (Galloway, 2009; Van Graan et al., 2016; Wang et al., 2013; Larue et al., 2015; National League for Nursing, 2015).

Health professionals need to possess critical thinking skills and be competent independent practitioners, who are able to manage varying disease conditions from a diverse patient population in a cross spectrum of healthcare settings (Larue et al., 2015; Rodriguez, 2013; McCabe, 2016; NLN, 2015). Simulation has therefore been considered a possible solution to the recorded disproportion between students, clinical instructors, and clinical placements (Rodriguez, 2013 & McCabe, 2016). This is because simulation possesses the ability to meet the educational

needs of health sciences students by facilitating a learner-centered, constructive, experiential and safe environment for developing the knowledge, skills and attitude required for different clinical setting and population (Aebersold & Tschannen, 2013; Galloway, 2009; McGrath et al., 2012; National League for Nursing, 2015).

2.4 The use of simulation in health sciences professions and its benefits

Since the adoption of simulation by the health sciences professions, research reports have revealed positive outcomes with simulation use, such as increased knowledge and skill competency, increased confidence and satisfaction of learners, with a recorded decreased risk of patient outcomes (Stunden et al., 2015; Ohtake & Erdley, 2013). Similarly, instructors have identified simulation as a standardized way to promote teamwork, collaboration, problem-solving, decision making, and critical thinking in a non-threatening environment among learners (Larue et al., 2015; Naude, 2016). Learners are also able to perform tasks and observe the outcome of the tasks carried out and they are given the chance to reflect on their experience and to engage in discussions about the learning experience at the debriefing phase (Rothgeb, 2008).

The following section will explore the use of simulation as it may be applied within the different departments which make up The School which is the setting for this study.

2.4.1 Simulation and Occupational Therapy education

Although the use of simulation in occupational therapy (OT) education is still growing, positive results have been documented. Research conducted by Shea (2015) & Ozelie et al. (2016) in their study that explored the use of simulation in the undergraduate OT curriculum at Samuel Merritt University (SMU) and a private university showed that the level of confidence in OT students increased in preparation for practice with the adoption of high fidelity simulation, improving critical thinking skills and helping them learn inter-professionally. Shea (2015) & Ozelie et al. (2016) recommend that institutions embrace SBE as it provides learners with opportunities for experiential learning, integrating theory and practice, improving their knowledge and communication skills.

2.4.2 Simulation and Physiotherapy education

Unlike in occupational therapy education, simulation is not regarded as a new concept in physiotherapy, as physiotherapists have often trained students on basic skills such as airway suctioning and hyper-inflation techniques using plastic mannequins with positive reports of critical thinking and decision-making skills developed by learners. The physiotherapy profession adopted the use of simulation to skills teaching and learning for the same reason as other health professions, because of the ethical and organizational issues such as patient safety and constrained clinical sites which are associated with the traditional method of clinical training (Blackstock & Gwendolen, 2007).

The use of high fidelity simulation (HFS) in physiotherapy was also used to expose undergraduate learners to complex cases and allow them a safe environment to engage in the patient's condition and plan the required physiotherapy intervention, during which facilitators could focus on the learners and provide support, rather than focusing on the patient. The adoption of simulation in physiotherapy therefore helped learners achieve learning outcomes and overcome a learning experience that is limited (Blackstock & Gwendolen, 2007).

Despite the benefits that simulation offered, the physiotherapists were concerned about ensuring realism of the environment, the ability to ensure the professional behavior of students to the artificial environment, and the transference of knowledge to the practice settings, with the greatest barrier being identified as the cost of equipment, space, staff and scenarios (Blackstock & Gwendolen, 2007; Jones & Sheppard, 2007). A study conducted by the Australian Catholic University (2017) showed the possibilities achievable with simulation by identifying physiotherapy competencies that can be developed using both LFS and HFS in physiotherapy with the successful transfer of skills to the clinical setting.

An essential competency expected of health science professionals in practice is to be able to work with other professionals through the exhibition of teamwork, communication and understanding of professional roles in patient care. As identified by Dennis et al. (2017), simulation successfully revealed these skills during an interprofessional simulated learning activity offered to physiotherapy and nursing students in their study.

11

2.4.3 Simulation and Pharmacy and Pharmacology education

In pharmacy and pharmacology education, improved pharmacotherapy knowledge, confidence in interpreting data, and clinical competence have been identified in learners after taking part in a simulation experience which involved the use of a human patient simulator to assist pharmacotherapy students understand the concept of dysrhythmias that seemed difficult when taught in class and cardiovascular medicine in general (Seybert, 2008; Naude, 2016). In the study conducted by Seybert (2008) and Naude (2016), observation sessions were organized for students to be able to see the changes in patient status with a right or a wrong medication administration. This helped prepare the students for future course assessments where they could apply problem-solving skills, with 10 out of the 15 group of students having a 100% pass with an average of > 95% across the group. Specific reference to nursing training in pharmacology with critically ill patients using HFS was made when after the simulation training, a reduced medication error rate (30.4% to 4%) was recorded in the study group, when the impact of training was evaluated in a real hospital environment (Branch, 2013).

2.4.4 Simulation and Nursing education

Simulation in Nursing education is well established (Sivertsen et al., 2016). Simulation was first introduced into nursing in 1911 when the first patient simulator manikin, "Mrs. Chase," was brought to the Hartford Hospital Training School for Nurses (National League for Nursing, 2015), and was used to facilitate experiential learning.

Similar to professions such as occupational therapy, physiotherapy and pharmacy and pharmacology, simulation provides nursing students with a safe environment to practice skills thereby building their confidence level and attainment of competence (Gonzales, 2017). Evidently, nursing participants in a South African qualitative study by Welman and Spies (2016) found that their students' encounter with simulation was beneficial as it helped to better equip them with the necessary skills and knowledge needed in a real clinical environment.

A study by Kim et al. (2016) and Wang et al. (2013) also showed that problem-based learning that is augmented with HFS was embraced by undergraduate nursing students more so than traditional lectures, with increased satisfaction being reported by the students.

2.4.5 Simulation and Sports Medicine

Engaging in sports and exercise often comes with setbacks such as sports injuries being incurred. The role of the sports health professional in such cases is vital as it is expected that a sports professional can identify the extent of the injury and the appropriate intervention required. There are sports conditions where a team approach is necessitated to provide the best patient care, meaning that the sport practitioner should be able to function in a team effectively. These abilities or skills of critical thinking, decision making, problem-solving and functioning in a team are often facilitated in a problem-based learning environment and have been shown to be possible with simulation-based learning (Rothgeb, 2008; Comfort & Abrahamson, 2010).

Currently, there is a dearth of information indicating the underuse of simulation in sports medicine, despite the potential for simulation to assist with achieving the competencies required of the sports health professional.

2.5 Challenges of implementing simulation

In contrast to the identified simulation benefits, Horsley and Wambach (2015) and Gudayu (2015) highlight certain shortcomings with the use of simulation. The researchers reported increased student anxiety during assessment due to instructor presence and claimed that the number of students with increased self-efficacy, confidence, and satisfaction with simulation experience was limited due to insufficient instructor assistance during simulation skills teaching. In pharmacy and pharmacology education, claims of students' limited exposure to simulation experience prior to assessment and being unclear of their roles during the simulation experience were made (Branch, 2013; Eukel et al., 2014). A similar complaint of inadequate theoretical preparation and orientation to the simulation environment was expressed by undergraduate nursing students in the qualitative study by Welman and Spies (2016), which made the students uncomfortable and unsure of the learning expectations.

Setting up a simulation laboratory for use is projected as being expensive as it includes the initial outlay, equipment maintenance, the need for technology experts, getting the needed number of persons to role-play interdisciplinary team members in a simulated learning activity, faculty development and continuing faculty/administrative/technical support. The expense and the lack

of technical support and knowledge about the simulation equipment have been identified as limiting factors for health professions like nursing, occupational therapy and physiotherapy to adopting simulation (Blackstock & Gwendolen, 2007; Jones & Sheppard, 2007; Rothgeb, 2008; Al-Ghareeb & Cooper, 2015; Li, 2017).

As observed by Rothgeb (2008), Kim et al. (2016) and Li (2017), institutions planning to set up a simulation laboratory only come to the realization that they lack the ability to use the simulation equipment, have no idea of the level of learners that would benefit from the experience, are unable to integrate simulation into the curriculum and are unable to enhance the reflective abilities of learners after the selection and purchase of the equipment.

Faculty are faced with fears related to technology, time constraints, and how to incorporate simulation into the nursing curriculum which has resulted in challenges in using the equipment and the reasons why equipment remains underutilized (Burns, 2008; Williams et al., 2016). According to Williams et al. (2016) nurse educators specifically, baby boomers, not technically proficient as other generational groups have been tagged to be faced with the challenge of integrating the use of high fidelity human patient simulators into the curriculum to meet the learning needs and styles of millennials.

As shown that simulation implementation requires educators to be technologically savvy and possess the know-how, the limited number of staff to champion SBE, who possess the knowledge and the skills needed to use simulation as a teaching and assessment tool, further increases the barrier to the adequate use of simulation (Lazzara et al., 2014; Nehring et al., 2013).

The perceptions of staff revealed that prior the introduction of simulation as a teaching methodology, staff were not consulted, felt pressured into using simulation, perceived it was only introduced as the new buzz with institutions aiming to be a part of the new trend and not because the need for simulation had been established which contributes to a low staff uptake of simulation (Miller & Bull, 2013).

Kim et al. (2016) therefore recommends that facilitators should focus on the learning outcomes of a simulation activity rather than on the provision of varied equipment and supplies, and as pointed out by Shea et al. (2014), new programs or practices in organizations often fail because leaders do not ensure sufficient institutional readiness for change. Institutional readiness refers to

the adoption of an organizational culture and policy, ensuring the availability of skilled personnel, structure, finance, and support to prepare staff psychologically and behaviorally to implement a change (Society of Hospital Medicine, 2016; Weiner et al., 2009).

The researcher, therefore, infers that for the benefits of simulation to be properly harnessed health sciences institutions need to ensure staff readiness for the adoption of SBE and requires all members of staff at the institution to tap into the initiative. This is supported by Leighton & Foisy-Doll (2016) who state that staff members need to be adequately prepared and should work together as a team to achieve successful student outcomes.

The researcher believes that the challenges encountered are aspects of simulation that should be properly understood and handled collectively by both the management and staff to ensure readiness for the initiative with subsequent successful implementation. This will be addressed in the next section below.

2.6 Readiness factors for simulation adoption

Due to the challenges encountered by institutions adopting the use of simulation, vital factors have been identified to help guide administrators to plan for the successful implementation of simulation programs and ensuring readiness for the adoption of simulation.

2.6.1 Science

The science of simulation is the hallmark of a simulation experience. This involves ensuring the intended SBE that the institution plans to provide is aligned with the vision, mission, goals and learning objectives of the institution's program. It refers to proper planning and preparation for the simulation experience by ensuring the learning objectives and scenarios used are aligned with the expected learner competencies and ensuring appropriate evaluation tools are designed (Lazzara et al., 2014; Beaubien & Baker, 2004; Galloway, 2009; Motola et al., 2013).

2.6.2 Staff

According to Lazzara et al. (2014) and Williams et al. (2016), a health education institution's workforce is its most important asset and its most expensive outlay. Due to the high cost of staff recruitment, many institutions have reduced their human resources. In addition to this, the staff's

existing workload, the lack of training of faculty members, and poor communication of SBE objectives gives rise to staff resisting the implementation of SBE.

Rothgeb (2008) indicated that the lack of experience of faculty members with the simulation equipment and a lack of training in SBE are the reasons for faculty's negative feelings towards simulation use. Faculty expressed the need for a user guide, training, support and time to plan for simulation, as they felt that their confidence with equipment use will increase with recurrent involvement.

It is therefore important to recruit interested faculty members as simulation champions to promote simulation, to strengthen educators as facilitators, and to implement incentive programs for integrating simulation (Lazzara et al., 2014 & Williams et al., 2016). As stated by Faz et al. (2014), to ensure nurse educators' buy-in, provision should be made for scenario observation in an environment that is non-threatening with an opportunity to ask questions pertaining to the simulation pedagogy. Online training has also been pinpointed as an effective approach to equipping academics (Kim et al., 2017).

2.6.3 Space and Supplies

The availability of a facility or location for SBE and training is vital for the successful implementation of simulation. This could be a dedicated space within a department or outside its walls. Planning for renovation or reconstruction of such a space is based on a needs assessment and should be done proactively. Setting up the designated simulation laboratory is highly expensive as it entails the design and restructuring of rooms, procurement of resources, and equipment. (Rothgeb, 2008; Lazzara et al. 2014). Although this is beyond the scope of the staff, it is important for management to ensure the availability and accessibility of space and supplies as motivating factors that the staff will work with for successful outcomes of simulation use.

2.6.4 Support

Institutional support and high-level management investment is a vital factor in the success of training program initiatives such as SBE. This entails seeking mutual partnerships with individuals who can help in the planning and execution of simulations, as SBE design and delivery require cooperation between technical and audiovisual professionals, experienced

persons in developing learning contexts, and content specialists. The support from these professionals will help ensure the training design is sound for educational instruction and that content is accurate and appropriate (Motola et al., 2013; Lazzara et al. 2014).

2.6.5 Success and Sustainability

Success and sustainability look at evaluating and sharing success stories throughout the institution, which is aimed at maintenance, not just development. It involves measuring pre-set outcomes which will help determine the efficiency of the program and help with future considerations (Motola et al., 2013; Lazzara et al. 2014). Such practices help ensure continuous success stories. This can be facilitated by ensuring there is a committee consisting of key stakeholders to steer the process.

These above-mentioned factors are in line with Seropian et al. (2004a) 8-steps recommendation cited by Rothgeb (2008) on how to establish a successful simulation program:

- Ensuring a vision is developed of the goal, those involved, and the way the laboratory will be used.
- Creating a plan to sketch initial and annual monetary requirements.
- Gaining support from stakeholders.
- The simulation setting construction should be, as per the vision and the plan, including the purchase of equipment.
- Training of all individuals anticipated to be involved.
- Curriculum development and,
- Determining policies and procedures.

2.7 CONCEPTUAL FRAMEWORKS FOR READINESS

A conceptual framework validated by Sharma et al. (2014) through evaluating the organizational readiness of early childhood education settings to implement nutrition and physical activity programs among children, was adapted in this study, by the researcher, to further explain the readiness factors needed to ensure successful implementation of simulation programs. The

framework theorizes that institutional factors and individual factors specific to the program's objectives, jointly contribute towards a person's readiness to implement a program effectively.

The conceptual framework highlights the three main antecedents for readiness that are connected to implementing a new program successfully. These factors are structural and external factors, staff attributes, and other psychological factors. The structural and external factors operate at the organizational level only, while staff attributes and other psychological factors are linked to both the organization and the individuals. It is said that organizational factors influence individual factors, and all these factors are theorized to collectively inform organizational readiness, which in turn influences program implementation. This is represented in Fig 2.1.

Similar to the Sharma et al. (2014) framework is that of Taplay et al. (2015) who discussed in their study the seven-phase process of adopting and incorporating simulation as a teaching strategy. The seven phases include securing resources; leaders working in tandem; getting it out of the box; learning about simulation and its potential for teaching; finding a fit; trialing of the equipment; and integrating into the curriculum. Elements of both theories are discussed in the sections below.



Fig 2.1 Conceptual framework for readiness (adapted from Sharma et al., 2014)

2.7.1 Structural and external factors

The structural and external factors are concepts measured at the organizational level of the framework. They are factors that influence the functioning of an institution and the institution's readiness to change. These factors include resources, policies, professional growth, training, and communication (Sharma et al., 2014). "Resources" include finance, infrastructure, and human resources, which includes the teaching and administrative staff members for the simulation program. This relates to Taplay et al. (2015) first phase of adoption which requires securing resources such as funds and making available a space for simulation set-up. "Policies" refers to the existence of written institutional policies on SBE.

"Professional growth and training" relates to making available training for staff to learn about simulation, which is in line with the process of "learning about simulation and its potential for teaching" identified by Taplay et al. (2015) and "communication" refers to the way institutions communicate information about simulation to their staff (Sharma et al., 2014).

2.7.2 Staff attributes

Sharma et al. (2014) defined staff attributes as attitudinal concepts that could influence the implementation of simulation at the individual and organizational levels. Staff attributes are therefore measured at both the organizational level and at the individual level. The staff attributes address staff cohesion, stress, staff authority, openness to change, clarity of goals and self-efficacy.

"Staff cohesion" refers to how the staff of the institution work together as a team. In Taplay et al. (2015) view, there is a joint effort from both the administrator and the simulation leader, which is referred to as leaders working in tandem. This is aimed at reducing the stress of workload. "Stress" is operationally described as staff members feeling strained and fatigued, which may result in staff resistance and can therefore impede simulation implementation. "Workload" has been identified as a cause of stress in the interpretation of the framework. "Staff authority" is viewed as the opportunity given by the management to staff for the implementation of innovative thoughts and changes.

"Openness to change" mirrors the overall attitude and readiness of the institution and its staff to the adoption of simulation. This requires a shift in thinking requiring staff to believe that the adoption of simulation and integrating into the curriculum is a viable choice for the benefit of students' learning (Sharma et al., 2014; Taplay et al., 2015). "Clarity of goals" shows how well the staff view the objectives of the new simulation program as appropriate with the institution's overall objectives. This relates to "finding a fit" in the proposition made by Taplay et al. (2015). The trialing of equipment is also one of the seven-phase process of adopting simulation which entails getting acquainted with the equipment through practice which gives rise to self-efficacy (Taplay et al., 2015). "Self-efficacy" according to Sharma et al. (2014) therefore measures capacity and ability of the institution and staff to implement simulation.
2.7.3 Other psychological factors

Psychological factors are referred to as the individual beliefs and attitudes influencing the acceptance and support for any change in the institution. These factors which include motivation, trust in leadership and perceived authority, are primarily measured at the individual level with the focus on lecturers involved in the implementation of simulation. "Motivation to change" is expressed as the staff member's perception of need (that is, do they value simulation?), the time required to make the change (is there release time for simulation), and pressure to implement simulation. These factors are seen to be vital in motivating the staff in "getting the equipment out of the box" for use (Sharma et al., 2014; Taplay et al., 2015). "Trust in leadership" is the extent of belief staff members have in the choice made by management to adopt simulation. Finally, "perceived authority" is the staff members' perception of their own authority in decision making in the institution (Sharma et al., 2014).

The framework is therefore adopted as it clearly supports the notion that readiness for change should be collective (institutional and individual) and should be assessed before embarking on the implementation of a program, as this will aid the prompt recognition of possible barriers to execution. The elements of this framework discussed are also reflected in the items on the instrument used for this study thus linking the elements of the study to the framework.

2.8 Conclusion

In this chapter, literature surrounding simulation and preparing for its use through adequate staff involvement has been explored. The next chapter will discuss the research methodology used in this study.

CHAPTER THREE

RESEARCH METHODOLOGY

3.1 Introduction

This section addresses the research design, research setting, population, sampling technique, data collection method, instrument, validity and reliability, data analysis and ethical considerations.

3.2 Research design

The research method shows how the research was carried out and the approach through which the research questions were answered (Brink et al., 2015). It involves a dense description of the research design. The research design is described as the steps taken by the researcher to answer the research question. It details the plan for gathering, measuring, and analyzing data. (De Vaus, 2001 & Brink et al., 2015).

A quantitative cross-sectional descriptive design was used in this study, as data was collected from different participants at a specific point in time within a limited duration (Olsen & George, 2004; Brink et al., 2015). Due to the dearth of information regarding staff readiness for simulation and limited research done to assess the readiness of staff for simulation within medical education, a **descriptive design** was chosen. This design is used where there is need for information regarding a phenomenon and helps in providing a picture of the concept as it occurs naturally (Brink et al., 2015).

3.3 Research setting

The research setting is described by Brink et al. (2015:59) as 'a specific place or places where the data are collected'. The choice of setting is dependent on the research question. The School, situated within the university, and comprised of the Departments of Nursing Education,

Occupational Therapy, Physiotherapy, Pharmacy and Pharmacology and the Centre for Exercise Science and Sports Medicine (CESSM), was the selected setting for the research, as there is an established simulation laboratory with potential participants being the stakeholders in any simulation development in The School.

3.4 Population and sample

The population of the study were the individuals who had met the criteria that was of interest to the researcher (Brink et al., 2015). The potential participants in this study were all employed as either full-time or part-time lecturers in the setting. The population was further delineated by applying the following inclusion and exclusion criteria.

3.4.1 Inclusion criteria

All participants had to be full-time or part-time lecturers with at least a 50% post and involved in the teaching of undergraduate students from the five departments within The School of the University.

3.4.2 Exclusion criteria:

- Lecturers who only teach postgraduate students in the five departments.
- Lecturers on a contract of less than 50%

These set of individuals were excluded as the study sought to explore *undergraduate* lecturers' readiness for simulation, and those on a contract less than 50% are assumed to have limited knowledge about the activities in their department that could help answer the research question.

3.4.3 Sampling technique

Sampling is the process of choosing individuals from a population of interest (Williams, 2006 & Brink et al., 2015). A non-probability sampling technique was chosen for the study as it permits the selection of persons knowledgeable about the phenomenon being investigated (Brink et al., 2015).

A total population sampling method was adopted which is a type of purposive sampling technique that entails choosing to examine the total population that possess a particular set of characteristics (Laerd, 2012). The researcher chose to study the entire population of lecturers employed in a 50% or greater post at the school of Therapeutic Health Sciences, as the size of the population that had the set of characteristics of interest, was very small.

3.4.4 Estimated sample size

The total sample size was reached by adding up the total number of lecturers who met the inclusion criteria from the five departments. The number of lecturers from each department that made up the sample are presented below:

Nursing (16 lecturers), Physiotherapy (13 lecturers), Pharmacy and Pharmacology (25 lecturers), Occupational Therapy (14 lecturers), Center for Exercise Science & Sports Medicine (4 lecturers). A total of 72 lecturers.

3.5 Data collection procedure

Permission to conduct the study was sought from the University Ethics Committee and the Postgraduate Assessors Committee. Once the required permissions had been obtained, the researcher introduced herself to the Heads of the five departments and explained the purpose of the study to them, requesting their permission to contact their full-time and part-time undergraduate lecturers and inviting them to participate in the study.

The lecturers identified were contacted electronically through email to explain the study, the data collection method and time required to complete the survey, after which the survey (Appendix E) was electronically sent using the REDCap[®] software. When a low response rate was observed, a meeting was arranged for the delivery of a self-administered questionnaire, to encourage participation. This was to facilitate a higher participation rate and ensure maximum data collection.

Confidentiality and anonymity was assured as no lecturer's name was required, only the departmental specialty.

3.5.1 Instrument

Wilkinson & Birmingham (2003) describe a research instrument as a tool for collecting data that is of interest to the researcher. The Simulation Culture Organizational Readiness Survey (SCORS) (Appendix F) instrument was used in data collection. It is a 24-item, 5-point Likert scale with scores ranging from 1 = none at all to 5 = very much, with higher scores indicating a higher level of readiness for simulation. The SCORS had four main sections:

Section A: Defined Need and Support for Change

Section B: Readiness for Culture Change

Section C: Time, Personnel, and Resource Readiness

Section D: Sustainability Practices to Embed Culture

The possible range of scores from the SCORS instrument is 36 to 180 and helps in determining the extent of readiness:

0---36 shows "Not ready"

37--72 shows "A little ready"

73-108 shows "Somewhat ready" 109-144shows "Moderately ready"145-180 shows "Very much ready".

The SCORS instrument was a modification of an instrument first developed by Fineout-Overholt & Menlnyk (2006) to assess organizational readiness. Leighton & Foisy-Doll (2016) adopted the instrument specifically to measure organizational readiness for SBE. Permission was obtained for its use from the authors (Leighton & Foisy-Doll, 2016). A demographic section was included in the survey to help identify responses from the five selected departments which would reflect on the extent of readiness for simulation for each department. The position of some questions was changed to help with imputation unto the REDCap[®] software. Instead of having questions 19, 20, 21 and 22 in ascending order in *"section c"*, the questions appeared in the following order: 20, 21, 19 and 22.

3.5.2 Validity and reliability

The instrument was validated by the original author (Fineout-Overholt & Menlnyk, 2006) using an expert panel of simulation educators and researchers and was further tested by Leighton & Foisy-Doll (2018). Fineout-Overholt and Menlnyk (2006) search to evaluate "Organizational Culture & Readiness for System-Wide Integration of Evidence-based Practice" states that both face and content validity of the instrument was established with internal consistency reliabilities >.85. The results of the test conducted by Leighton & Foisy-Doll (2018) also proves the instrument to be reliable with an internal consistency of .96.

3.6 Data analysis

Descriptive statistics were used to describe and summarize data using tabulated description (tables), and statistical comments (discussion of the results).

This statistical analysis was used, as the study adopted a non-experimental descriptive design, with the aim of exploring and describing the phenomena and the ability to identify problems with current practice or justify their practice (Brinks et al., 2015).

Due to the close range of the results gathered from the five departments within The School, inferential statistics were also used to test if there was a significant difference in the results.

The four sections of the survey instrument were analyzed to determine the extent of readiness as per the four sections of the SCORS questionnaire viz:

- a) Defined need and support for change,
- b) readiness for culture change,
- c) time, personnel and resource readiness and
- d) sustainability practices to embed culture.

The facilitating and limiting factors were determined based on the items reflected in the SCORS questionnaire and supported by a discussion of findings and an in-depth literature review.

3.7 Ethical considerations

Brink et al. (2015) clearly states that the onus lies on the researcher to ensure the research is conducted ethically from the phase of conceptualization until the research findings are disseminated.

3.7.1 Permission to conduct the study

The proposal was presented to the Department of Nursing Education for peer review after which the necessary permissions were sought and obtained to gain permission to proceed with the study.

Permission to conduct the study was sought and obtained from the following:

- The University Postgraduate Assessor's Committee (Appendix A).
- The University's Human Research Ethics Committee (Medical), Protocol number: M170238 (Appendix B).
- The Heads of Department of the five departments to gain access to the lecturers.
- Permission to use the SCORS instrument was obtained from the original authors (Appendix C).

3.7.2 Informed consent

Voluntary informed consent was established as outlined by Brink et al. (2015): to ensure that the potential participants fully understand what is expected of them and are then able to decide if they choose to participate.

• In this study, the prospective participants were full-time and part-time undergraduate lecturers in the five departments within The School. They had a good comprehension of the information given by the researcher as it was related to them in English, which is the common language in the selected university.

- The information was conveyed in written form to all the potential participants by the researcher via electronic (email) this included: the information document and invitation to participate (Appendix D).
- A clear detailed description of the purpose of the study, the proposed plan for sending out the surveys were explained to all participants prior to them consenting to participate.
- Participants were informed of the nature and benefits of the research and notifying them that there would be no risk to themselves if they chose to volunteer.
- The potential participants were given an opportunity to ask any questions about the study to the researcher by making available the email addresses and contact numbers of the researcher and her supervisors.
- Participants were informed of their right to withdraw from the research without any adverse consequences to themselves.

3.7.3 Confidentiality and Anonymity

Confidentiality refers to the researcher being responsible for keeping safe the details of participants during the research, and anonymity refers to namelessness and keeping participants characteristics a secret (Brink et al., 2015).

- Confidentiality and anonymity were ensured as participants' names were not included on the survey, only names of the department where they were currently lecturing at the time of data collection, was identified in the resulting research report.
- Data gathered from participants were kept safe with only the researcher and her supervisors being able to access the data.

3.7.4 Non-maleficence

Non-maleficence refers to the participant's right to protection from any discomfort and harm (Burns & Grove, 2001).

• Participants were assured there was no harm in taking part in the study and could withdraw at any time of the data collection period, with no penalty to themselves.

3.7.5 Justice

Ensuring justice in research refers to the right to participants being selected and treated fairly. Participants ought to be selected based on the criteria relating to the research problem and not because of their availability (Burns & Grove, 2001; Brink et al., 2015).

- Participants were selected based on the formal inclusion and exclusion criteria for this study.
- Participation was voluntary.
- All participants were asked the same questions and every participant's opinion was regarded as of equal importance.

3.7.6 Respect

This refers to the participant's right to self-determination, which must be maintained by ensuring voluntary consent, permission to withdraw and avoiding coercion on behalf of the researcher (Burns & Grove, 2001; Brink et al., 2015).

- Participants were informed of their right to choose to voluntarily participate in the study.
- Participants were treated with respect, during all the interaction with the researcher.

3.8 Conclusion

This chapter provided an overview of the research design and data collection methods as used in this study. The data analysis and interpretations of the findings from the study are presented in the next chapter.

CHAPTER FOUR

DATA ANALYSIS AND RESULTS

4.1 Introduction

This chapter presents the findings of this research. The data that was gathered was analyzed based on the objectives of the research which were:

- To determine the extent of readiness of lecturers from the five departments to integrate simulation-based education in The School.
- To identify/determine factors that will prevent or promote the full utilization of simulation in The School.

The survey instrument was divided into two parts. The first part elicited the demographic information of the respondents while the second part, which consists of the four sections of the SCORS instrument, addressed the objectives of this research as stated above. The possible range of scores from the SCORS instrument is 36 to 180 and determined the extent of readiness of the School for simulation: 0 to 36 shows "**Not ready**", 37 to 72 shows "**A little ready**", 73-108 shows

"Somewhat ready", 109-144 shows "Moderately ready", 145-180 shows "Very much ready".

The guidebook which accompanied the SCORS instrument provided guidance on how the results gathered from the instrument used were interpreted. The participants' responses were registered on a 5-point Likert scale. The mean of the scores for each item on the instrument was calculated to identify the factors that promoted the use of simulation within The School, as well as the factors that prevented The School from being fully ready to use simulation. As determined by the SCORS guidebook, a mean score of 3 and above is favorable while a mean score below 3 indicates area to be focused on by each department or The School.

The results gathered from the study show that findings from the five departments were within close range. This led to the need to use inferential statistics to test if there was a significant difference in the readiness for simulation amongst the five departments within The School. An

analysis of variance (ANOVA) and Tukey test was used to determine the significance in the results and difference in readiness. According to Andale (2017), ANOVA test is used to determine the overall significance of the results while the Tukey test will help identify where differences lie and which specific department's means when compared, are different. The research results are presented below.

4.2 Response rate

Table 4.1 presents a summary of the respondents to the distributed survey instrument. With the use of the REDCap[®] software, 72 surveys were sent out. Of the 72 surveys sent out, 50 completed surveys were returned and analyzed. A response rate of 69.4% was achieved in this study. According to Mundy (2002), an acceptable response rate for a population survey is 60%. This shows that this study had a good response rate.

Table 4.1 Respondents to the distributed survey

Number of surveys distributed	72
Returned surveys	50
Questionnaires used in analysis	50
Percentage of actual respondents	69.4%

4.3 Demographic information

The respondents were requested to provide information on their age, gender, department, employment status and years of working in their respective departments.

4.3.1 Age distribution

Table 4.2 provides an overview of the age distribution of the respondents, with 19 (38%) respondents in the age group 31 to 40 years, 10 (20%) respondents were in the age group 41 to 50 years, and 9 (18%) in the age range of 20 to 30. Only 6 (10%) respondents each were in the age groups 51 to 60, and older than 60.

Table 4.2 Age group of particip	pants
---------------------------------	-------

Age group	Frequency (n= 50)	Percentage (%)
20-30	9	18
31-40	19	38
41-50	10	20
51-60	6	12
>60	6	12

4.3.2 Gender

Of the 50 respondents presented in Table 4.3, 41 (82 %) were female and 9 (18%) were male.

Table 4.3 Gender of respondents

Gender	Frequency (n=50)	Percentage (%)
Male	9	18
Female	41	82
Total	50	100

4.3.3 Department

Table 4.4 shows the number of respondents from the five departments. It is indicated that 15 (93.75%) of the respondents were from the department of nursing education while 11 (44%) of the respondents were from the department of pharmacy and pharmacology.

Department	Frequency of surveys	Frequency of surveys returned	Percentage of the surveys
	distributed (n=72)	(n=50)	returned/distributed
			(%)
Nursing	16	15	93.75
Occupational therapy	14	10	71.43
Physiotherapy	13	10	76.92
Pharmacy and	25	11	44
pharmacology			
CESSM	4	4	100

Table 4.4 Number of respondents from each of the five departments

4.3.4 Employment status

From the available data shown in Table 4.5, 11 (22%) were part-time undergraduate lecturers (employed on a 50% or more post) while majority, 39 (78%) of the respondents were full-time undergraduate lecturers.

Table 4.5 Employment statu	s of respondents
----------------------------	------------------

Employment status	Frequency (n=50)	Percentage (%)
Full-time	39	78
Part-time	11	22

4.3.5 Years of service of respondents in their departments

Table 4.6 shows that the majority of the 50 respondents appeared to have between 1 year to 10 years of working experience, of which 13 (26%) were in the range of 1 to 5 years and 13 (26%) in the group of 6 to 10 years.

Years of service	Frequency (n=50)	Percentage (%)
1-5	13	26
6-10	13	26
11-15	8	16
16-20	4	8
21-25	5	10
> 25	7	14

Table 4.6 Years of service of respondents

4.4 Readiness of lecturers within The School

The School had a score of 107.52 which showed that The School is "somewhat ready" (73-108) for simulation.

4.4.1 The readiness of lecturers in the five departments within The School

Fig 4.1 shows that the departments of nursing (109.9), physiotherapy (112.8) and pharmacy

and pharmacology (113.8) are "moderately ready" for simulation, and while the department of occupational therapy (106.2) is "somewhat ready", the center for exercise science and sports medicine (71.5) is "a little ready" for simulation.



Fig 4.1 Departmental readiness for simulation

4.5 Factors preventing or promoting the readiness of lecturers to use simulation within the School.

The results pertaining to this part of the research are presented according to the four sections of the SCORS instrument and are analyzed with regards to the demographic variables (department, age, gender, years of service and employment status).

4.5.1 Section A: Defined need and support for change within The School.

As indicated in Table 4.7 below, the lack of adequate communication of a clear strategic vision for SBE (mean=2.78) (see question 3), the lack of provision of a written commitment to SBE (mean=2.34) (see question 4), and insufficient provision of funds to support the commitment to SBE (mean=2.42) (see question 5) were factors which prevented the lecturers within The School from being fully ready for simulation. The School having innovation, experiential learning and quality student experiences clearly described as central to the mission and philosophy of the institution (mean=4.02) (see question 1), having clearly defined the need to consider SBE integration (mean=3.64) (see question 2) and educators articulating a need for SBE integration into the curriculum (mean=3.5) (see question 8) suggest there was a defined need and support for change which promoted their readiness to use simulation.

S/n	SECTION A: DEFINED NEED AND SUPPORT FOR CHANGE	Item
		mean
1	To what extent are innovation, experiential learning and quality student experiences clearly described as central to the mission and philosophy of your institution?	4.02
2	To what extent has your department clearly defined the need to consider SBE integration?	3.64
3	To what extent have administrators within your department communicated a clear strategic vision for SBE?	2.78
4	To what extent have administrators within your department provided a written commitment to SBE?	2.34
5	To what extent have administrators within your department provided funding to support the commitment to SBE?	2.42
6	To what extent does your department promote the need for SBE based on current evidence, standards, and guidelines?	3.18
7	To what extent is SBE currently being used as a teaching modality in your department?	3.06
8	To what extent have the educators you work with articulated a need for SBE integration into the curriculum?	3.5
9	To what extent have the educators in your department verbalized a commitment to SBE integration into the curriculum?	3.38

Table 4.7 Defined need and support for change within The School.

4.5.2 Section B: Readiness for culture change within The School

Table 4.8, below, shows that the limiting factors for simulation readiness were poor availability of credentialed or trained simulationists who could mentor/coach others (mean = 2.54), in question 12; the inadequate numbers of individuals who model SBE best practice (mean = 2.58), in question 13; as well as graduate-level prepared researchers available to assist in research to develop new knowledge, appropriate to each department's mission (mean = 2.78), in question 15.

The lecturers within The School have a strong belief that it is the right time to implement simulation as indicated in question 18, with a mean of 4.18. This factor, alongside the proficiency of lecturers in the use of technology (mean = 3.56) in question 14 are both enabling factors for simulation.

Table 4.8 Readiness for culture change within The School.

	SECTION B: READINESS FOR CULTURE CHANGE	Item
		Mean
		(5)
10	To what extent is there a critical mass of professionals who already possess strong SBE:	
a	Knowledge	3.02
b	Skills	2.74
c	Positive Attitudes	3.5
11	To what extent do administrators support culture change including the efforts required to implement and sustain SBE program integration?	3.12
12	To what extent are there credentialed or trained simulationists who mentor/coach others, including, other simulationists?	2.54
13	To what extent does your department have individuals who model SBE best practice?	2.58
14	To what extent are staff/faculty proficient in the use of technology? (i.e. computer systems, AV and IT systems)	3.56
15	To what extent are there graduate level prepared researchers available to assist in research to develop new knowledge, appropriate to your department's mission?	2.78
16	To what extent are librarians available within your institution to help search for evidence based-practice and related simulation resources?	3.5
17	To what extent are your librarians accessed to search for evidence-based practice and related simulation resources?	3.12
18	To what extent do you believe that now is the right time to implement a culture change to support SBE?	4.18

4.5.3 Section C: Time, Personnel and Resource Readiness

Although the lecturers agreed that they have access to technology (mean = 4.04) in question 20 and are provided with support with technology use, in question 21 with a mean of 3.98, there is an insufficiency with human resources (simulation personnel) (mean = 2.3), release time for educators to lead the integration of SBE (mean = 2.2), development of physical learning spaces (2.82) and equipment (mean = 2.7) in questions 19a, 19b, 19c, 19d and 19e, all of which impact on the readiness of lecturers to use simulation within The school.

Table 4.9 Time, personnel and resource readiness within The School.

	SECTION C: TIME, PERSONNEL AND RESOURCE READINESS	Item Mean
		(5)
20	To what extent do employees in your department have access to quality technology, including computers, audiovisual equipment, and other institutional technologies?	4.04
21	To what extent is support available to learn and manage technologies that support education?	3.98
19	To what extent are fiscal resources available to support SBE in the following areas:	
a	Human resources (simulation personnel)?	2.3
b	Education?	2.52
c	Release time to lead integration of SBE?	2.2
d	Development of physical learning spaces?	2.82
e	Equipment?	2.7
22	To what extent are there existing simulation champions (people who will go the	
	extra mile to advance simulation) in the current environment among:	
a	Administrators?	2.24
b	Clinicians?	2.22
c	Educators?	2.84
d	Technology Specialists?	2.48
e	Administrative Assistants and Support Staff?	1.88

4.5.4 Section D: Sustainability practices to embed culture within The School.

The section on "*sustainability practices*" indicated that decisions regarding SBE influenced by clinicians (mean = 2.54) in question 24a and administration (mean = 2.74) in question 24c was limited. A promoting factor was the fact that measurement and sharing of outcomes is part of the culture of The School. This is presented in table 4.10 below.

Table 4.10 Sustainability practices to embed culture within The School.

	SECTION D: SUSTAINABILITY PRACTICES TO EMBED CULTURE	Item
		Mean
		(5)
23	To what extent is the measurement and sharing of outcomes part of the culture of the	3.34
	department in which you work?	
24	To what extent are decisions regarding SBE influenced by:	
a	Clinicians?	2.54
b	Educators?	3.72
c	Administration?	2.74

4.5.5 Readiness of lecturers within The School by sections

Table 4.11 shows that the section with the highest need of intervention is "Section c", which requires time, personnel and resources to facilitate lecturers' readiness to use simulation (2.69).

Table 4.11 An overview of readiness of lecturers within The School by sections

SCORS QUESTIONNAIRE	SCHOOL'S SECTION MEAN
Section A: Defined need and support for change	3.15
Section B: Readiness for culture change	3.15
Section C: Time, personnel and resource readiness	2.69
Section D: Sustainable practices to embed culture	3.09

4.6 Factors preventing or promoting the readiness of lecturers to use simulation within each department of The School.

4.6.1 Section A: Defined need and support for change by department

All the five departments as indicated in Table 4.12 below, appear to be insufficiently aware of a written commitment to SBE and seem to require more funds to support the commitment to SBE (questions 4 and 5). The center for exercise science and sports medicine is the only department with the need for the greatest intervention as they only recognize that innovation and experiential learning is central to their mission and philosophy (mean = 3.5) as indicated in question 1.

S/n	SECTION A: DEFINED NEED AND SUPPORT FOR	ITEM MEAN					
	CHANGE	(5)					
		NUR	OT	PHY	PHA	CESSM	
1.	To what extent are innovation, experiential learning and	4.13	4.5	3.5	4.09	3.5	
	quality student experiences clearly described as central to						
	the mission and philosophy of your institution?						
2.	To what extent has your department clearly defined the	3.93	3.7	3.9	3.64	1.75	
	need to consider SBE integration?						
3.	To what extent have administrators within your	3	2.8	3.1	2.7	1.3	
	department communicated a clear strategic vision for						
	SBE?			-			
4.	To what extent have administrators within your	2.33	2.7	2.6	2.18	1.25	
	department provided a written commitment to SBE?						
5.	To what extent have administrators within your	2.93	2.5	2.3	2.18	1.25	
	department provided funding to support the commitment						
	to SBE?	0.45	0.5				
6.	To what extent does your department promote the need	3.47	3.6	3.4	2.82	1.5	
	for SBE based on current evidence, standards, and						
-		2.4	0.0	27	2.26	1.05	
1.	To what extent is SBE currently being used as a teaching	3.4	2.3	3.7	3.36	1.25	
0	To what award house the advantage you work with	2.0	27	26	2.5	1.5	
δ.	10 what extent have the educators you work with	3.8	3.7	3.0	3.5	1.5	
0	articulated a need for SBE integration into the curriculum?		2.0	2.6	2 1 0	1.05	
9.	10 what extent have the educators in your department	3.67	5.8	3.0	3.18	1.25	
	verbalized a commitment to SBE integration into the						

Table 4.12 Defined need and support for change by department

4.6.2 Section B: Readiness for culture change by department

All the departments except pharmacy and pharmacology (mean = 3, see question 10b), lack professionals who already possess strong SBE skills. In question 10c, the positive attitudes of lecturers and lecturers' proficiency with the use of technology in question 14, facilitated the readiness of lecturers for simulation. Table 4.13 highlights these factors.

	SECTION B: READINESS FOR CULTURE	ITEM MEAN (5)				
	CHANGE	NUR	ОТ	PHY	PHA	CESSM
10.	To what extent is there a critical mass of professionals who already possess strong SBE:		1	I		I
a	Knowledge	3.27	2.8	2.8	3.36	2.25
b	Skills	2.87	2.8	2.6	3	1.75
c	Positive Attitudes	3.4	3.9	3.6	3.6	2.3
11	To what extent do administrators support culture change including the efforts required to implement and sustain SBE program integration?	2.8	3.3	3.6	3.45	1.75
12	To what extent are there credentialed or trained simulationists who mentor/coach others, including, other simulationists?	3.07	1.4	3.1	2.82	1.25
13	To what extent does your department have individuals who model SBE best practice?	3.13	1.8	3	2.64	1.25
14	To what extent are staff/faculty proficient in the use of technology? (I.e. computer systems, AV and IT systems)	3.67	3.7	3.5	3.64	2.75
15	To what extent are there graduate level prepared researchers available to assist in research to develop new knowledge, appropriate to your department's mission?	2.87	2.5	3	3	2
16	To what extent are librarians available within your institution to help search for evidence-based practice and related simulation resources?	3.53	3.7	3.6	3.55	2.5
17	To what extent are your librarians accessed to search for evidence-based practice and related simulation resources?	3	3	3	3	2
18	To what extent do you believe that now is the right time to implement a culture change to support SBE?	4.6	4.4	4.2	4.18	2

Table 4.13 Readiness for culture change by department

4.6.3 Section C: Time, personnel and resource readiness by department

In Table 4.14, question 19 shows that all the departments have insufficient fiscal resources, but nursing (mean = 3.07) and physiotherapy (mean = 3.1) in possession of equipment for SBE, and only nursing having physical learning space for simulation (mean = 3.4). Simulation champions' existence in the five departments is limited with only the departments of physiotherapy (mean = 3.1) and pharmacy and pharmacology (mean = 3.18) having sufficient educators to champion simulation.

In this section, CESSM have access to technology (mean = 3.25) and are provided with support to learn and manage technologies that support education (mean = 4), just like other departments.

	SECTION C: TIME, PERSONNEL AND RESOURCE	ITEM MEAN				
	READINESS	(5)				
		NUR	OT	PHY	PHA	CESSM
20	To what extent do employees in your department have access	3.73	4.5	4.2	4.18	3.25
	to quality technology, including computers, audiovisual					
	equipment, and other institutional technologies?					
21	To what extent is support available to learn and manage	3.4	4.4	3.9	4.5	4
	technologies that support education?					
19	To what extent are fiscal resources available to support					
	SBE in the following areas:			-		
а	Human resources (simulation personnel)?	2.33	2.4	2.5	2.36	1.25
b	Education?	2.67	2.5	2.6	2.55	1.75
c	Release time to lead integration of SBE?	2	2	2.4	2.64	1.75
d	Development of physical learning spaces?	3.4	2.6	2.7	2.55	2.25
e	Equipment?	3.07	2.4	3.1	2.64	1.25
22	To what extent are there existing simulation champions					
	(people who will go the extra mile to advance simulation) in					
	the current environment among:					
а	Administrators?	2	2	2.7	2.55	1.75
b	Clinicians?	2.27	1.6	2.6	2.45	2
c	Educators?	2.93	2.4	3.1	3.18	2
d	Technology Specialists?	2.27	2.8	2.7	2.64	1.5
e	Administrative Assistants and Support Staff?	1.47	1.5	2.2	2.55	1.75

Table 4.14 Time, personnel and resource readiness by department

4.6.4 Section D: Sustainability practices to embed culture by department

Measurement and sharing of outcomes of simulation practices is being carried out in all departments except for nursing (mean = 2.87) which fell below expectation. Educators are the major individuals influencing the decisions regarding SBE in nursing, occupational therapy, physiotherapy and pharmacy and pharmacology, except for the center for exercise science and sports medicine (mean = 2.75). It is only pharmacy and pharmacology that involves clinicians (3.55) and administrators (3.36) in decisions regarding SBE. This is presented in Table 4.15 below.

 Table 4.15 Sustainability practices to embed culture by department

	SECTION D: SUSTAINABLE PRACTICES TO EMBED	Item Mean				
	CULTURE	(5)				
		NUR	OT	PHY	PHA	CESSM
23	To what extent is the measurement and sharing of outcomes part of the culture of the department in which you work?	2.87	3.2	3.7	3.73	3.5
24	To what extent are decisions regarding SBE influenced by:					
a	Clinicians?	2.33	2.1	2.2	3.55	2.5
b	Educators?	3.8	3.6	3.7	4.09	2.75
c	Administration?	2.47	2.9	2.4	3.36	2.5

4.6.5 Sectional readiness of lecturers by department

As shown below in Table 4.16, all the five departments require the provision of time, personnel and resources to be ready for simulation use. While the department of nursing and occupational therapy need to adopt sustainability practices, the center for exercise science and sports medicine require intervention in all sections to ensure they are ready for simulation.

Table 4.16 Sectional readiness of lecturers in specific departments

SCORS QUESTIONNAIRE	DEPARTMENTAL SECTION MEAN					
	NUR	ОТ	PHY	PHA	CESSM	
Section A: Defined need and support for	3.41	3.29	3.3	3.08	1.61	
change						
Section B: Readiness for culture change	3.29	3.06	3.31	3.33	1.93	
Section C: Time, personnel and resource	2.63	2.59	2.89	2.89	2.04	
readiness						
Section D: Sustainable practices to embed	2.87	2.95	3	3.68	2.81	
culture						

4.6.6 The difference in the readiness for simulation in the five departments

From the ANOVA results as shown in Table 4.17, there was a statistically significant difference (p-value=0.0346) in readiness among the different departments.

	Df	SS	MS	F-value	p-value
Department	4	6005	1501.2	2.847	0.0346
Residuals	45	23726	527.2		

Table 4.17 Difference in the readiness for simulation in the five departments

The results of a Tukey test indicated that there were statistically significant readiness differences between nursing department and CESSM (p-value=0.036), CESSM and physiotherapy department (p-value=0.031) and between CESSM and pharmacy and pharmacology department (p-value=0.023).

The results indicate that there were no statistically significant differences in readiness values between nursing education and occupational therapy (p-value=0.994), nursing education and physiotherapy (0.997), nursing education and pharmacy and pharmacology (0.992), physiotherapy and occupational therapy (0.967), pharmacy and pharmacology and occupational therapy (0.94), pharmacy and pharmacology and physiotherapy (0.994), certain therapy (0.94), pharmacy and pharmacology and physiotherapy (0.999). CESSM is therefore the outlier.

4.7 Factors preventing or promoting the readiness of lecturers to use simulation by age group.

Across the four sections of the SCORS instrument, factors identified to be impacting on readiness to use simulation by age group is the same as those factors indicated and interpreted in the sections above. Importantly, as shown in Table 4.18 (question 14) and Table 4.19 (questions 20 & 21) below, lecturers have access to technology, were provided with support to manage technology, and were found to be technologically proficient across all age group.

	SECTION B: READINESS FOR CULTURE CHANGE	Item Mean (5)				
		20-30	31-40	41-50	51-60	>60
10.	To what extent is there a critical mass of professionals who already possess strong SBE:					
a	Knowledge	3.33	2.89	2.6	3	3.67
b	Skills	3	2.68	2.4	2.67	3.17
c	Positive Attitudes	3.78	3.26	3.3	3.5	4.17
11	To what extent do administrators support culture change including the efforts required to implement and sustain SBE program integration?	3.22	3.16	2.8	2.83	3.67
12	To what extent are there credentialed or trained simulationists who mentor/coach others, including, other simulationists?	2.33	2.79	2.4	2.17	2.67
13	To what extent does your department have individuals who model SBE best practice?	2.89	2.47	2.2	2.33	3.33
14	To what extent are staff/faculty proficient in the use of technology? (I.e. computer systems, AV and IT systems)	3.78	3.63	3	3.33	4.17
15	To what extent are there graduate level prepared researchers available to assist in research to develop new knowledge, appropriate to your department's mission?	3.22	2.74	2.3	2.67	3.17
16	To what extent are librarians available within your institution to help search for evidence-based practice and related simulation resources?	3.44	3.26	3.6	4	3.67
17	To what extent are your librarians accessed to search for evidence- based practice and related simulation resources?	2.89	3.21	3	3.33	3.17
18	To what extent do you believe that now is the right time to implement a culture change to support SBE?	4.44	4.21	3.8	4.5	4

Table 4.18 Readiness for culture change by age group

Table 4.19 Tin	e, personnel	and resource	readiness b	by age group
----------------	--------------	--------------	-------------	--------------

	SECTION C: TIME, PERSONNEL AND RESOURCE READINESS	Item Mean				
		20-30	31-40	41-50	51-60	>60
20	To what extent do employees in your department have access to quality technology, including computers, audiovisual equipment, and other institutional technologies?	4.33	4.11	3.9	3.83	3.83
21	To what extent is support available to learn and manage technologies that support education?	4.11	4.11	3.9	3.67	3.83
19	To what extent are fiscal resources available to support SBE in the following areas:					_
a	Human resources (simulation personnel)?	2.33	2.37	1.9	2.67	2.33
b	Education?	2.33	2.74	2.4	2.17	2.67
c	Release time to lead integration of SBE?	2.11	2.37	1.9	2.5	2
d	Development of physical learning spaces?	3.11	2.95	2.4	2.5	3
e	Equipment?	2.67	2.68	2.5	2.5	3.33
22	To what extent are there existing simulation champions (people who will go the extra mile to advance simulation) in the current environment among:					·
a	Administrators?	2.56	2.32	2	2.17	2
b	Clinicians?	2.56	2.11	1.8	2.33	2.67
c	Educators?	2.89	2.79	2.3	3.17	3.5
d	Technology Specialists?	2.67	2.79	2	1.83	2.67
e	Administrative Assistants and Support Staff?	2.33	1.89	1.5	1.67	2

4.7.1 Sectional readiness by age group

Table 4.20 shows that readiness for culture change is lowest in the age group of 41-50 and highest in the age group >60. In addition, across all the age group, a need for time, personnel and resource has been identified.

Table 4.20 Sectional readiness by age group

SCORS	Section mean by age group						
	20-30	31-40	41-50	51-60	>60		
Section A: Defined need and support for change	3.14	3.22	3	3.04	3.3		
Section B: Readiness for culture change	3.3	3.12	2.85	3.12	3.53		
Section C: Time, personnel and resource readiness	2.83	2.77	2.38	2.58	2.82		
Section D: Sustainable practices to embed culture	3.17	3	2.88	3.33	3.33		

4.7.2 The difference in the readiness for simulation by age group

The p-value (0.7) for the ANOVA test shows that there is no statistically significant difference among the different age categories regarding readiness.

4.8 Factors preventing or promoting the readiness of lecturers to use simulation by employment status.

Across the four sections, an important point to note in "Section b" as presented in Table 4.21 is that in questions 10a and 10b, full-time lecturers expressed a lack of sufficient knowledge (mean = 2.9) and skills (mean = 2.69) to use simulation as opposed to part-time lecturers. These are limiting factors to use simulation.

	SECTION B: READINESS FOR CULTURE CHANGE	Item M	lean (5)
		FT	РТ
10.	To what extent is there a critical mass of professionals who already		
	possess strong SBE:		
a	Knowledge	2.9	3.45
b	Skills	2.64	3.09
c	Positive Attitudes	3.41	3.82
11	To what extent do administrators support culture change including the	3.05	3.36
	efforts required to implement and sustain SBE program integration?		
12	To what extent are there credentialed or trained simulationists who	2.49	2.73
	mentor/coach others, including, other simulationists?		
13	To what extent does your department have individuals who model SBE	2.51	2.82
	best practice?		
14	To what extent are staff/faculty proficient in the use of technology?	3.54	3.64
	(I.e.		
	computer systems, AV and IT systems)		
15	To what extent are there graduate level prepared researchers available	2.79	2.73
	to assist in research to develop new knowledge, appropriate to your		
	department's mission?		
16	To what extent are librarians available within your institution to help	3.38	3.91
	search for evidence-based practice and related simulation resources?		
17	To what extent are your librarians accessed to search for evidence-	3.05	3.36
	based practice and related simulation resources?		
18	To what extent do you believe that now is the right time to implement	4.21	4.09
	a culture change to support SBE?		

Table 4.21 Readiness for culture change by employment status

4.8.1 Sectional readiness by employment status

Table 4.22 shows that part-time lecturers had a more positive view of The school's readiness for simulation.

i dolo 1.22 Doollon loddinobb o' ompio imone blacab	Table 4.22 Sect	tion readiness	by emp	lovment status
---	-----------------	----------------	--------	----------------

SCORS	Section mean by employment status			
	FT	РТ		
Section A: Defined need and support for change	3.1	3.31		
Section B: Readiness for culture change	3.09	3.36		
Section C: Time, personnel and resource readiness	2.65	2.83		
Section D: Sustainable practices to embed culture	2.95	3.7		

4.8.2 The difference in the readiness for simulation by employment status

The study found out that there is no statistically significance difference in readiness between fulltime and part-time workers (t-value=-1.0079, df=12.889, p-value=0.3321). That is, the average readiness for full-time workers (105.2564) is not statistically different from the average readiness for part-time workers (115.5455).

4.9 Sectional readiness by years of experience

Table 4.23 shows that lecturers who have worked between 6 and 10 years within The School have indicated that all sections require attention to facilitate readiness for simulation use.

Table 4.23 Sectional readiness by years of experience

SCORS		Sectional mean by years of experience						
	1-5	6-10	11-15	16-20	21-25	>25		
Section A: Defined need and support for change		2.79	3.6	3.67	3.09	3		
Section B: Readiness for culture change		2.8	3.44	3.18	3.16	3.35		
Section C: Time, personnel and resource readiness		2.54	2.51	2.38	3.05	2.56		
Section D: Sustainable practices to embed culture		2.81	2.75	2.44	3.85	3.43		

4.9.1 The difference in the readiness for simulation by years of experience

The p-value (0.693) for the ANOVA test shows that there is no statistically significant difference among the different years of experience regarding readiness.

4.10 Conclusion

This chapter presented the findings of data collected from 50 respondents who are lecturers in the five departments within the School of the University. The chapter began by discussing the response rate and the sequence in which the results were analyzed and would be presented. Research findings were then presented in tabular and written form. The discussion of the results presented will be addressed in the next chapter.

CHAPTER FIVE

DISCUSSION OF FINDINGS, CONCLUSION, AND RECOMMENDATIONS

5.1 INTRODUCTION

In the previous chapter, the findings of the study were stated. In this chapter, the researcher will discuss the findings, and the conclusions drawn from the findings of the research conducted. The recommendations arising, and the limitations of the study will also be stated in this chapter.

5.2 DISCUSSION OF FINDINGS

The purpose of this research was "to examine the readiness of lecturers from the five departments within The School where the research took place to adapting simulation-based education and to identify factors preventing or promoting the successful use of the new methodology". The findings of this study revealed the extent of readiness of The School and its five departments. The research results also revealed areas that promoted the lecturers' readiness to use simulation and the factors that prevented The School from being fully ready for simulation.

The following discussion of the SCORS findings in this study, on the factors that promoted the lecturers' readiness to use simulation and the factors that prevented The school from being fully ready for simulation are guided by elements of the framework, the SCORS instrument and readiness factors for simulation adoption identified in reviewed literature.

5.2.1 Readiness of the school to integrate SBE

The SCORS instrument used in this study helped determine the extent of readiness of The School to integrate SBE. The instrument had four sections which elicited responses to various factors that could determine readiness for SBE. According to the results, The School had an overall score of 107.52 which showed that The School is "somewhat ready" for SBE but not fully ready.

Within The School, five departments were surveyed and yielded varied scores showing they fared differently on their readiness for SBE. While the departments of nursing (109.9), physiotherapy (112.8) and pharmacy and pharmacology (113.8) were "moderately ready" for SBE, occupational therapy (106.2) was "somewhat ready" for SBE and CESSM (71.5) was "a little ready" for SBE.

As per the sections on the SCORS instrument, it was revealed that The School requires more time, personnel and resources to be ready for simulation. The department of nursing and occupational therapy need to adopt sustainability practices for simulation, with the center for exercise science and sports medicine requiring intervention in all sections to ensure they are ready for simulation.

For an organization or school to be institutionally ready refers to the adoption of an organizational culture and policy, ensuring the availability of skilled personnel, structure, finance and support to prepare staff psychologically and behaviorally to be able to implement a change (Society of Hospital Medicine, 2016; Weiner et al., 2009). This description of institutional readiness can be linked to Sharma et al. (2014) framework of readiness where certain factors such as structural and external factors; staff attributes; and psychological factors are grouped as antecedents to readiness and successful implementation for change.

5.2.2 Antecedents of readiness supporting SBE

5.2.2.1 Structural and external factors

Lazzara et al. (2014) states that the hallmark of a simulation experience is the "science" behind it, which is explained as ensuring the intended SBE that the institution plans to provide is aligned with the vision, mission, goals and learning objectives of the institution's program which is tagged as "policies" in Sharma et al. (2014) framework. Based on the results of this study, having innovation, experiential learning and quality student experiences central to the mission and philosophy of the institution, prepares lecturers to use simulation, as these are educational qualities that are inherent in SBE.

Employees across the five departments in The School had access to quality technology, including computers, audiovisual equipment, and other institutional technologies and were provided with support to learn and manage technologies that support education. These are promoting factors for readiness to expand their education methods to include simulation.

5.2.2.2 Staff attributes

Sharma et al. (2014) theorize that staff attributes such as their "openness to change", and "self-efficacy", are antecedents to readiness. "Openness to change", mirror the overall attitude and readiness of the institution and staff to adopt simulation, while "self-efficacy," looks at the capacity

and ability of the institution and staff to implement simulation. Based on the attribute of being open to change, this study's results show that educators verbalized a commitment to SBE integration into the curriculum, indicating there is a value to change and a shift in thinking to incorporating initiatives which facilitates the readiness of staff for simulation (Taplay et al., 2015).

A category of simulation (high fidelity simulation) which is computerized and is currently adopted by most institutions requires technological skills which have often been cited as a barrier to simulation use, as lecturers, especially baby boomers, often express that they are not technologically savvy (Evans, 2017; Williams et al., 2016). In contrast, lecturers within The School, indicated technology proficiency across all age groups, and are ready to embark on the use of high fidelity simulation.

Staff authority according to Sharma et al. (2014) is the opportunity given by the management to staff for the implementation of innovative thoughts and changes. In this study, educators had considerable influence in decisions regarding SBE within The School as revealed in the research results. The authority given serves to trigger change readiness in the staff for the implementation of simulation.

5.2.2.3 Psychological factors

As identified from the study's results, The School had a defined need to consider SBE integration which motivates the staff to embrace the initiative, with the potential for a successful implementation of SBE. This is supported by the conceptual framework adapted from Sharma et al. (2014) and Taplay et al. (2015) that highlights that having a perceived need motivates staff to get the equipment out of the box for use which is a solution to the issue of under-use of simulation equipment.

5.2.3 Antecedents to readiness hindering SBE

5.2.3.1 Structural and external factors

Within The School, the lack of adequate communication of a strategic vision and written commitment for simulation, limited the lecturers' use of simulation. "Communication" is, as operationally defined in the framework by Sharma et al. (2014), an antecedent to readiness, and refers to the way institutions communicate information about simulation to their staff. The School's poor communication of a clear strategic vision for SBE was a factor identified to have prevented the lecturers from being fully ready to use simulation.

Adequate funding is a vital factor identified by Lazzara et al. (2014) for readiness and successful implementation of an initiative like SBE. As seen from the research results, the insufficient provision of funds to support the commitment to SBE prevented the lecturers within The School from being fully ready to implement simulation.

Asides from resources referring to availability of funds, it also refers to the availability of skilled personnel, and according to the results, the limiting factor for simulation readiness is the poor availability of credentialed or trained simulationists who could mentor/coach others. The inadequate numbers of individuals who model SBE best practice as well as graduate-level prepared researchers available to assist in research to develop new knowledge, appropriate to each department's mission, hinders the uptake of simulation. The presence of these personnel within The School would help to ensure there is a well thought out plan to balance the workload of interested lecturers wanting to introduce simulation, thereby reducing stress which can hinder the staff from embracing the change to SBE (Sharma et al., 2014; Taplay et al., 2015; Lazzara et al., 2014; Williams et al., 2016).

The study's results revealed that some of the staff in The School, except for the department of pharmacy and pharmacology were knowledgeable, but had limited SBE skills. It is also interesting to note that full-time lecturers expressed a lack of sufficient knowledge and skills to use simulation as opposed to part-time lecturers. These are limiting factors to use simulation which therefore hints at the need for professional growth and training to equip the staff for the successful implementation of SBE (Faz et al., 2014; Sharma et al., 2014, Kim et al., 2017).

To ensure total readiness for simulation, it is therefore important to ensure lecturers are skilled to use simulation. This would require attention to be paid to the specialist training of credentialed simulationists to provide support to simulation users. The skills of lecturers to use simulation is very important as it directly impacts on the clinical learning of students. These skills entail the ability to plan, implement, evaluate the simulation experience, and the debriefing of students, post simulation. When planning for simulation, facilitators need to ensure learners are orientated to the simulation environment and are informed of their roles prior to the simulation experience. The facilitator is also expected to adopt a debriefing style that best suits the situation in which the simulation experience had been organized and be able to ascertain that learning had occurred through the development of reflective thinking practice in learners (Beaubien & Barker, 2004).

The lack of these skills was identified in a study by Branch (2013) & Eukel et al. (2014), where students were unsure of their roles during the simulation experience due to limited prior exposure to simulation before assessment. The inability of facilitators to enhance reflective abilities of learners during the debriefing phase of simulation has been highlighted (Kim et al., 2016 & Li, 2017). The negative impact of the staff of The School's lack of SBE and debriefing skills on students' learning therefore requires that adequate steps be taken to ensure lecturers are skilled to facilitate simulation-based learning as well as ensuring the availability of support to effectively implement simulation practices (Al-ghareeb & Cooper, 2015).

5.2.3.2 Staff attributes

To ensure the simulation training design is sound for educational instruction, it is recommended that support from experts are sought in decisions regarding SBE (Motola et al., 2013; Lazzara et al., 2014). Lazzara et al. (2014), suggest that partnerships with individuals who can help in the planning and execution of simulations should be sought, as SBE design and delivery requires cooperation between technical and audio-visual professionals, experienced persons in developing learning contexts, in conjunction with experts in the course content area. A professional team approach ensures simulation training is instructionally sound and that content is accurate and appropriate to help achieve successful student outcomes.

The results from this study indicate that within The School, the major decision makers are educators with only pharmacy and pharmacology involving clinicians and administrators in decisions regarding SBE. This according to Sharma et al. (2014) shows there is a strain in "staff cohesion" within the other departments in decisions regarding SBE.

5.2.3.3 Psychological factors

"Perceived time" is a motivating factor of readiness of staff for SBE (Sharma et al., 2014). Based on the result, release time for educators to lead the integration of SBE was perceived to be insufficient and has been identified as one of the factors that prevented the lecturers from being fully ready for simulation.

According to Lazzara et al. (2014), planning towards establishing successful outcomes with SBE is important. This entails ensuring the success and sustainability of programs which are inclusive of measuring and sharing of outcomes and collaborating with experts regarding the creation and

implementation of the simulation experience. Despite nursing being the profession in the health sciences to first introduce simulation as a teaching pedagogy (National League for Nursing, 2015), the nursing department in this study fell short in measuring and sharing their outcomes in comparison to the other departments included in the study. The researcher believes this lack of measurement of simulation outcomes could impede the opportunity to identify the potential of SBE in helping students transfer skills to practice, as well as preventing further growth of the pedagogy through the dissemination of outcomes and student results.

5.3 Limitation

The setting of this study was one selected school, where the participants were lecturers from the five departments (Nursing, Occupational therapy, Physiotherapy, Pharmacy and pharmacology, and the Center for Exercise Science and Sports Medicine). In addition, although a total sample was used, the sample size was relatively small. Therefore, the findings of this study cannot be generalized to the readiness of lecturers in a different context. A wider survey of the entire faculty would assist in this regard, although The School does provide for its own needs at present with regards to simulation resources.

While anonymity was guaranteed, the fact that the research was confined to one School may have led to participants believing their responses may be identifiable, even if only by means of the department where they work. As a result, there is a possibility that this may have led to social desirability bias (Smith, 2007).

A further limitation may have been attribution bias (Carroll, 2014), either due to respondents trying to justify or blame issues on their own or other people's behaviors, or even have been self-serving bias (Boyes, 2013), if they thought that certain responses may improve their access to simulation in the future.

5.4 Conclusion

The research question that guided this study was, "What is the extent of readiness of lecturers from the five departments within The School at the selected University for the use of SBE?"

With limited research having been done in the South African context on readiness factors for simulation, the researcher embarked on this research in an endeavor to investigate the extent of
readiness of lecturers within one school in a university to use simulation, and identifying factors influencing their readiness to use simulation. From the research findings, a perceived need and time for simulation, the availability of skilled teaching and administrative staff, educational opportunities for staff, and collaboration with experts facilitated by management should be planned for when preparing to adopt the use of simulation as these are some of the factors that could facilitate or hinder readiness for SBE.

A major aspect of simulation adoption is having a strategic vision and mission statement towards simulation. As it may seem difficult to change the university's vision and mission statement, the vision and mission statement towards simulation can be put into The School's policy document and be communicated to the members of staff.

The proposed clinical training model in South Africa clearly stipulates that 30% of nursing students' learning be dedicated to theoretical learning, while 70% be directed towards practical aspects of learning. Out of the 70%, 40% is for role taking in the clinical setting and 30% for experiential learning, of which, 20% is for simulation (The Nursing Education Stakeholders (NES) Group, 2012). To ensure role taking (where learners become socialized to the profession and function as part of the health care team) of learners in clinical settings, learners are prepared in skills and simulation laboratory to obtain skills needed in the practice setting (The Nursing Education Stakeholders (NES) Group, 2012). As recommended by the proposed clinical training model, it is observed that simulation takes a large percentage of experiential learning for learners' preparation for clinicals. This makes it important for adequate investment to be made towards simulation in The School.

The issue of funding has also been raised as an impacting readiness factor in this study which is not a direct responsibility of lecturers but of higher level management. A major source of funding could be obtained from the clinical grants for clinical training released by the National Department of Health (NDOH), to increase SBE. Similarly, institutions of learning have a part to play to motivate for the clinical training grant funding to be spent on simulation. This entails the involvement of stakeholders at management level with the capacity to develop policies, plan towards simulation and drive funding processes. It also involves facilitators recording and measuring positive student outcomes with simulation use which will serve as evidence to motivate stakeholders to advocate for funds and to gain the government's buy-in for SBE. From literature reviewed, faculty's fear of technology, with specific reference to baby boomers not being technologically savvy, was identified as a reason for the underuse of high fidelity simulation (Williams et al., 2016). In this study, it is interesting to point out that lecturers, including baby boomers, were technologically proficient, which is vital for the use of high fidelity simulation like patient simulators.

It is also important to note that simulation does not have to be expensive as the major focus should be the objective of the simulation and choosing the type of simulation that best meets the objective (Kim et al., 2016). The misconception of simulation being equated with high tech equipment needs to be cleared up through training. This is because simulation is not restricted to the use of equipment, but it also includes role-playing and the use of standardized patients.

This research has shown that The School is somewhat ready for simulation and the readiness varied from department to department. The center for exercise and sports medicine (CESSM), which was one of the departments assessed in this study, was found to be "a little ready" for simulation based on the results. When the results gathered from CESSM department were compared with those of other departments, CESSM was the only department with a statistically significant difference in results. From anecdotal evidence, CESSM was the only department that has not started using the simulation laboratory despite the benefits it offers. The result is therefore, a true reflection of the department and serves as a valuable measure that can help guide the department to focus on areas of need when they plan to embark on simulation use.

5.5 Recommendations

It is therefore recommended that to initiate simulation and improve The School's degree of readiness, The School does the following:

Evaluate the readiness of their lecturers: This will help identify areas of need that can be addressed to facilitate a successful simulation experience for both the learners and facilitators and all participants involved. The SCORS instrument used in this study is a valuable tool that can be adopted by institutions to evaluate their readiness, prior to the purchasing of equipment and the building of simulation centers. As this marks the first time of staff readiness for simulation being assessed within The School, it is recommended that the exercise be repeated upon acting on the areas of need of The School to help identify improvement on their degree of readiness.

This should not be a once-off event but a continuous one to ensure The School is fully ready and to ensure there is a sustenance of the positive result.

Involve staff in the planning for simulation: Upon the decision to adopt simulation, staff especially lecturers should be consulted about their views and need for simulation and how they perceive simulation-based learning fulfilling an educational gap and improving their student outcomes. Staff involvement through consultations will help promote staff commitment to the use of simulation, thereby limiting resistance and the underuse of resources through non-use of equipment and supplies. This aspect is one that is embraced by The School to an extent and has proven to facilitate The School's readiness for simulation.

Provide training to empower staff for the implementation of the initiative: As revealed by the research results, skilled simulation personnel are limited within The School. It is therefore recommended that personnel are made available. The need for the presence of a simulation expert within the institution is valuable as he or she can provide guidance on the simulation initiative. Staff members who are interested in simulation should also be identified and work as a team with the simulationist. Interested members should be trained with opportunities to learn and manage simulation which can be provided for by management through financing conference or workshop attendance. On-line webinars should be promoted, and attendance encouraged, as this form of training does not require lecturers to be absent from their offices for an extended period of time. This set of interested staff can in turn act as simulation champions for The School. Simulation knowledge and skills acquisition by staff is essential to ensure facilitators can develop, implement and evaluate their own simulation-based experiences for successful clinical learning of students in preparation for practice.

As there is one simulation laboratory available within The School, with certain departments being less aware of what it offers, there is a need to break the silos of departmental learning activities and having resources being available to all departments and not only accessible to certain departments within the broader university. This also refers to inter-professional teaching and training where simulation successes can be shared and adopted. The employment of a simulation coordinator assists in reducing the silo mentality, ensuring the simulation resources within The School are enjoyed by all departments.

The different departments within the university should be made aware of new teaching and learning resources by advertising through editorials, news flashes, lunch time training sessions and demonstrations. Creating awareness informs the lecturers of what is available in the institution to be incorporated into teaching and learning.

Since simulation is taking a great part in the clinical learning of learners in this present day, it is important for The School adopting simulation to ensure adequate preparation of their staff and setting for simulation use. This will help ensure the full benefits of simulation are harnessed in equipping learners that are satisfied with their learning experience and fit for practice.

References

Aebersold, M. & Tschannen, D. (2013). "Simulation in Nursing Practice: The Impact on Patient Care". *OJIN*. 18(2), Manuscript 6.

Al-Elq, A.H. (2010). Simulation-based medical teaching and learning. *J Family Community Med*. 17(1), p35–40.

Al-Ghareeb, A.Z. & Cooper, S.J. (2015). Barriers and enablers to the use of high-fidelity patient simulation manikins in nurse education: an integrative review. *Nurse Educ. Today.* p1-6.

Andale (2017). *What is the Tukey Test / Honest Significant Difference?* Available at: http://www.statisticshowto.com/tukey-test-honest-significant-difference/. Accessed 23rd Oct 2017.

Australian Catholic University (2017). *Simulation learning for physiotherapy students*. Available at:

http://www.acu.edu.au/about_acu/faculties,_institutes_and_centres/health_sciences/about_the_fa culty/news_and_events/news/simulation_learning_for_physiotherapy_students. Accessed 1st Jun 2017.

Beaubien, J.M. & Baker, D.P. (2004). The use of simulation for training teamwork skills in healthcare: how low can you go? *Qual Saf Health Care*. 13, i51–i56.

Blackstock, F.C. & Gwendolen, J.A. (2007). High-fidelity patient simulation in physiotherapy education. *The Australian journal of physiotherapy*. 53(1), p3-5.

Boyes, A. (2013). The self-serving bias - definition, research, and antidotes. Available at:

https://www.psychologytoday.com/blog/in-practice/201301/the-self-serving-biasdefinitionresearch-and-antidotes. Accessed 30th Oct 2017.

Branch, C. (2013). Pharmacy Students' Learning and Satisfaction with High-Fidelity Simulation to Teach Drug-Induced Dyspepsia. *Am J Pharm Educ*. 77(2), p30.

Brink, H., van der Walt, C. & van Rensburg, G. (2015). Sampling. In: Ristic, D. *Fundamentals of Research Methodology for Healthcare Professionals*. 3rd ed. Cape Town: Juta & Company Ltd. p139-141.

Burns, H.K. (2008). Integrating Simulation into Nursing Curriculum. Available at:

https://pdfs.semanticscholar.org/55d9/94a91d50f41c2d5462677a592a99c13f98c5.pdf. Accessed 20th Apr 2017.

Burns, N. & Grove, S.K. (2001). *The practice of nursing research: conduct, critique and utilization*. Philadelphia: W.B. Saunders.

Business Dictionaries (2017). *Readiness*. Available at: http://www.businessdictionary.com/definition/readiness.html. Accessed 31st May 2017.

Carroll, R.T. (2014). *Attribution biases*. Available at: http://skepdic.com/attribution.html. Accessed 30th Oct 2017.

Comfort, P. & Abrahamson, E. (2010). *Sports rehabilitation and injury prevention*. 1st ed. UK: John Wiley & Sons, Ltd. p298-306.

Dennis, D., Furness, A., Duggan, R. & Critchett, S. (2017). An interprofessional simulationbased learning activity for nursing and physiotherapy students. *Clinical Simulation in Nursing*. 13(10), p501-510.

Dennis, D., Sainsbury, D., Redwood, T., Ng, L. & Furness, A. (2016). Introducing Simulation Based Learning Activities to Physiotherapy Course Curricula. *Creative Education*, **7**, p878-885. Available at: http://www.scirp.org/journal/PaperInformation.aspx?paperID=66497. Accessed 5th Dec 2016.

De Vaus, D.A. (2001). Research Design in Social Research. 1st ed. London: SAGE. p9.

Eukel, H.N., Frenzel, J.E., Skoy, E.T., Focken, R.L. & Fitz, A.L. (2014). An introductory pharmacy practice experience using simulated patient care activities in a pharmaceutical care laboratory environment. *Currents in Pharmacy Teaching and Learning*. 6, p682–691.

Evans (2017). *The Many Degrees of Fidelity in Simulation of Clinical Practice*. Available at: http://simulationfornurseeducators.blogspot.co.za/p/resources-and-tools-for-simulation.html. Accessed 31st May 2017.

Faz, B., Van Sell, S. & Sheriff, S. (2014). Simulation Teaching: Developing Instructor Confidence. *International Journal of Nursing*. 1(2), p49-63.

Fineout-Over-holt, E. & Melnyk, B.M. (2006). "Organizational Culture & Readiness for System-Wide Integration of Evidence-based Practice Survey". In: Fineout-Overholt, E. & Melnyk, B.M. (2011). *Evidence-based practice in nursing & healthcare: A guide to best practice*. 2nd Ed. Philadelphia, PA: Lippincott, Williams, & Wilkins.

Frotjold, A. (2015). *The introduction of high fidelity simulation learning into a preregistration nursing course: the lived experience of nurse academics*. Available at: http://researchdirect.westernsydney.edu.au/islandora/object/uws%3A36176/datastream/PDF/vie w. Accessed 3rd Mar 2017.

Galloway, S. J. (2009). "Simulation Techniques to Bridge the Gap Between Novice and Competent Healthcare Professionals". *OJIN*. 14(2), Manuscript 3.

Gonzales, N.J. (2017). Use of high fidelity simulation in teaching nursing skills: A phenomenological inquiry. *Imperial Journal of Interdisciplinary Research (IJIR)*. 3(5), p1456-1464.

Gudayu, T.W., Badi, B.M. & Asaye, M.M. (2015). Self-Efficacy, Learner Satisfaction, and Associated Factors of Simulation Based Education among Midwifery Students: A Cross-Sectional Study. *Education Research International*. 2015, p1-7.

Harder, N. (2010). Use of Simulation in Teaching and Learning in Health Sciences: A Systematic Review. *Journal of Nursing Education*. 49 (1), p23-28.

Horsley, T. L., & Wambach, K. (2015). Effect of nursing faculty presence on students' anxiety, self-confidence, and clinical performance during a clinical simulation experience. *Clinical Simulation in Nursing*. 11(1), p4-10.

Jeffries, P.R. (2005). A framework for designing, implementing, and evaluating simulations used as teaching strategies in nursing. *Nurs. Educ. Perspect.* 26, p96-103.

Jones, A. & Sheppard, L. (2007). Can human patient simulators be used in physiotherapy education? *The Internet Journal of Allied Health Sciences and Practice*. 5(2), p1-5.

Kenney, J.K. (2014). *The future of simulations in allied healthcare education and training: A modified delphi study identifying their instructional and technical feasibility*. Available at: http://ufdc.ufl.edu/UFE0046484/00001. Accessed 16th Jan 2017.

Ker, J. & Bradley, P. (2010). Simulation in medical education. In: Swanwick, T. *Understanding Medical Education: Evidence, Theory and Practice*. Association for the Study of Medical Education. p164-180. Available at: http://faculty.uoit.ca/kapralos/csci5530/kerBradley_2010.pdf. Accessed 28th Jun 2017.

Kim, J., Park, J.-H. & Shin, S. (2016). Effectiveness of simulation-based nursing education depending on fidelity: a meta-analysis. *BMC Medical Education*. 16, p152.

Kim, S., Park, C. & O'Rourke, J. (2017). Effectiveness of online simulation training: Measuring faculty knowledge, perceptions, and intention to adopt. *Nurse Education Today*. 51, p102–107.

Laerd dissertation. (2012). *Total population sampling*. Available at: http://dissertation.laerd.com/total-population-sampling.php. Accessed 26th Jan 2017.

Larue, C., Pepin, J. & Allard, E. (2015). Simulation in preparation or substitution for clinical placement: A systematic review of the literature. *Journal of Nursing Education and Practice*. 5(9), p132-140.

Lateef, F. (2010). Simulation-based learning: Just like the real thing. *J Emerg Trauma Shock*. 3(4), p348-352.

Lazzara, E.H., Benishek, L.E., Dietz, A.S., Salas, E. & Adriansen, D.J. (2014). Eight Critical Factors in Creating and Implementing a Successful Simulation Program. *The Joint Commission Journal on Quality and Patient Safety*. 40(1), p21-29.

Leighton, K. & Foisy-Doll, C. (2016). *Development and use of the Simulation Culture Organizational Readiness Survey (SCORS)*. Available at: http://www.nursinglibrary.org/vhl/handle/10755/603882. Accessed 1st Sep 2016.

Leighton, K., Foisy-Doll, C. & Gilbert, G.E. (2018). Development and psychometric evaluation of the simulation culture organizational readiness survey. *Nurse Educator*. 00(0), p1-5.

Li, S. (2017). *The Role of Simulation in Nursing Education: A Regulatory Perspective*. National Council of State Boards of Nursing. Available at: https://www.ncsbn.org/Suling2.pdf. Accessed 2nd Jun 2017.

Lopreiato, J.O., Downing, D., Gammon, W., Lioce, L., Sittner, B., Slot, V., Spain, A.E. & the Terminology & Concepts Working Group. (2016). *Healthcare Simulation Dictionary*. Rockville, MD: Agency for Healthcare Research and Quality. Available at: https://www.ahrq.gov/sites/default/files/wysiwyg/professionals/quality-patientsafety/patientsafety-resources/research/simulation_dictionary/sim-dictionary.pdf. Accessed 3rd Jul 2017.

McCabe, D.E., Gilmartin, M.J. & Goldsamt, L.A. (2016). Student self-confidence with clinical nursing competencies in a high-dose simulation clinical teaching model. *Journal of Nursing Education and Practice*. 6(8), p52-58.

McGrath, M., Lyng, C. & Hourican, S. (2012). From the simulation lab to the ward: Preparing 4th year nursing students for the role of staff nurse. *Clinical Simulation in Nursing*. 8(7), pe265-e272.

Miller, A. & Bull, B.M. (2013). Do you want to play? Factors influencing nurse academics' adoption of simulation in their teaching practices. *Nurse Education Today*. 33, p241–246.

Motola, I., Devine, L.A., Chung, H.S., Sullivan, J.E. & Issenberg, S.B. (2013). Simulation in healthcare education: A best evidence practical guide. AMEE Guide No. 82. *Medical Teacher*. 35, pe1511–e1530

Muller, M. (2009). *Nursing Dynamics*. 4th ed. Sandton: Heinemann Publishers (Pty) Ltd. p174257.

Mundy, D. (2002). *A question of response rate*. Science Editor. 25(1), p25-26. Available at: https://www.councilscienceeditors.org/wp-content/uploads/v25n1p025-026.pdf. Accessed 16th Apr 2018.

National League for Nursing. (2015). A Vision for Teaching with Simulation: A Living Document from the National League for Nursing NLN Board of Governors. Available at: http://www.nln.org/docs/default-source/about/nln-vision-series-(position-statements)/visionstatement-a-vision-for-teaching-with-simulation.pdf?sfvrsn=2. Accessed 20th Apr 2017.

Naude, A. (2016). *The opinions on and use of simulation in undergraduate pharmacy education at south African universities*. Available at: http://scholar.ufs.ac.za:8080/xmlui/bitstream/handle/11660/3330/NaudeA.pdf?sequence=1. Accessed 6th Jul 2017.

Nehring, W.M., Wexler, T., Hughes, F. & Greenwell, A. (2013). "Faculty Development for the Use of High-Fidelity Patient Simulation: A Systematic Review". *IJHSE*. 1(1), p1-34.

Ohtake, P. & Erdley, W. (2013). Does the type of simulated patient (Manikin vs Role player) impact knowledge, confidence, self-efficacy and satisfaction in a simulated critical care rehabilitation experience? *The Journal of the Society for Simulation in Healthcare*. 8(6).

Olsen, C. & George, D.M. (2004). *Cross-Sectional Study Design and Data Analysis*. Available at: http://www.collegeboard.com/prod_downloads/yes/4297_MODULE_05.pdf. Accessed 4th Jun 2017.

Oxford Dictionary (2017). *Readiness*. Available at:

https://en.oxforddictionaries.com/definition/readiness. Accessed 31st May 2017.

Ozelie, R., Both, C., Fricke, E. & Maddock, C. (2016). "High-Fidelity Simulation in Occupational Therapy Curriculum: Impact on Level II Fieldwork Performance". *The Open Journal of Occupational Therapy*. 4(4), p1-11.

Rodriguez, S.M. (2013). "The impact of limited clinical sites on pre-licensure nursing education programs: Current issues and recommendations for the future". Master of Arts in Nursing

Scholarly Projects. Paper 73. Available at:

http://sophia.stkate.edu/cgi/viewcontent.cgi?article=1072&context=ma_nursing. Accessed 6th Sep 2016.

Rothgeb, M.K. (2008). Creating a Nursing Simulation Laboratory: A Literature Review. *Journal* of Nursing Education. 47(11), p489-494.

Seybert, A.L., Kobulinsky, L.R. & McKaveney, T.P. (2008). Human Patient Simulation in a Pharmacotherapy Course. *American Journal of Pharmaceutical Education*. 72(2), p1-8.

Sharma, S.V., Upadhyaya, M., Schober, D.J. & Byrd-Williams, C. (2014). A Conceptual Framework for Organizational Readiness to Implement Nutrition and Physical Activity Programs in Early Childhood Education Settings. *Prev Chronic Dis.* 11, p190. Available at: https://www.cdc.gov/pcd/issues/2014/pdf/14_0166.pdf. Accessed 13th Jan 2017.

Shea, C. (2015). "High-Fidelity Simulation: A Tool for Occupational Therapy Education". *The Open Journal of Occupational Therapy*. 3(4), p1-12.

Shea, M.C., Jacobs, S.R., Esserman, D.A., Bruce, K. & Weiner, B.J. (2014). Organizational readiness for implementing change: a psychometric assessment of a new measure. *Implementation Science*. 9(7), p2.

Sivertsen, N., McNeill, L. & Muller, A. (2016). A redo station after debrief improves learning in undergraduate nursing simulation. *Clinical Simulation in Nursing*. 12(11), p469-472.

Smith, W.W. (2007). Social desirability bias and exit survey responses: The case of a first nations campground in Central Ontario, Canada. *Tourism Management*. 28(3), p917-919.

Society of Hospital Medicine. (2016). *Quality & Innovation: Institutional readiness*. Available at:

https://www.hospitalmedicine.org/Web/Quality_Innovation/Quality_101/Institutional_Engageme nt/Institutional_Readiness/Web/Quality__Innovation/Quality_101/Institutional_Engagement/In stitutional_Readiness.aspx?hkey=646d4187-a266-4b1f-8d11-47f717a25730. Accessed 6th Sep 2016.

67

Sowerby, H. (2015). *Recent graduates' perspective on the efficacy of nursing simulation laboratory experiences*. Available at:

https://scholarworks.waldenu.edu/cgi/viewcontent.cgi?referer=https://www.google.co.za/&httpsr edir=1&article=1554&context=dissertations. Accessed 3rd Aug 2016.

Stunden, A., Halcomb, E. & Jefferies, D. (2015). Tools to reduce first year nursing students' anxiety levels prior to undergoing objective structured clinical assessment (OSCA) and how this impacts on the student's experience of their first clinical placement. *Nurse Education Today*. 35, p987–991.

Taplay, K., Jack, S., Baxter, P., Eva, K. & Martin, L. (2015). The process of adopting and incorporating simulation into undergraduate nursing curricula: A grounded theory study. *Journal of Professional Nursing*. 31(1), p26–36.

The Nursing Education Stakeholders (NES) Group. (2012). A proposed model for clinical nursing education and training in South Africa. *Trends in Nursing*. 1(1), p1-12.

Van Graan, A.C., Williams, M.J. & Koen, M.P. (2016). Professional nurses' understanding of clinical judgement: A contextual inquiry. *Health SA Gesondheid*. 21, p280-293.

Wang, A.L., Fitzpatrick, J.J. & Petrini, M.A. (2013). Use of simulation among chinese nursing students. *Clinical Simulation in Nursing*. 9(8), pe311-e317.

Weiner, B.J., Lewis, M.A. & Linnan, L.A. (2009). Using organization theory to understand the determinants of effective implementation of worksite health promotion programs. *Health Education Research*, 24(2), p292–305.

Welman, A. & Spies, C. (2016). High fidelity simulation in nursing education: considerations for meaningful learning. *Trends in Nursing*. 3(1), p1-16.

Wilkinson, D. & Birmingham, P. (2003). Using Research Instruments: A Guide for Researchers.UK: Psychology Press. p3.

Williams, A.T., Finney, A. & Elliott, M. (2016). *Overcoming the barriers of integrating simulation into a nursing curriculum*. Available at:

http://www.xitheta.org/images/Simulation_Integration_ELM_Final1_24x36.pdf. Accessed 20th Apr 2017.

Williams, M.K. (2006). Sampling. Available at:

https://www.socialresearchmethods.net/kb/sampling.php. Accessed 4th Jun 2017.

APPENDIX A



Private Bag 3 Wits, 2050 Fax: 027117172119 Tel: 02711 7172076

Reference: Mrs Sandra Benn E-mail: <u>sandra.benn@wits.ac.za</u>

> 12 June 2017 Person No: 1392573 PAG

Miss TA Awogbemila Unit 33 21 Barnowl Street Amberview Centurion 0157 South Africa

Dear Miss Awogbemila

Master of Science in Nursing: Approval of Title

We have pleasure in advising that your proposal entitled Assessing staff readiness for simulation in a Health Sciences Institution has been approved. Please note that any amendments to this title have to be endorsed by the Faculty's higher degrees committee and formally approved.

Yours sincerely

Usen

Mrs Sandra Benn Faculty Registrar Faculty of Health Sciences

APPENDIX B



R14/49 Miss Tolulope Awogbernila

HUMAN RESEARCH ETHICS COMMITTEE (MEDICAL)

CLEARANCE CERTIFICATE NO. M170238

NAME:	Miss Tolulope Awogbemila
DEPARTMENT:	Nursing Education University of the Witwatersrand
PROJECT TITLE:	Assessing Staff Readiness for Simulation in a Health Sciences Institution
DATE CONSIDERED:	24/02/2017
DECISION:	Approved unconditionally
CONDITIONS:	Title Change (24/05/2017)
SUPERVISOR:	Hilary Thurling
APPROVED BY:	ellatofon
	Professor P. Cleaton-Jones, Chairperson, HREC (Medical)
DATE OF APPROVAL:	24/04/2017

This clearance certificate is valid for 5 years from date of approval. Extension may be applied for.

DECLARATION OF INVESTIGATORS

To be completed in duplicate and **ONE COPY** returned to the Research Office Secretary in Room 10004.10th floor, Senate House/3rd floor, Phillip Tobias Building, Parktown, University of the Witwatersrand. I/We fully understand the conditions under which I am/we are authorised to carry out the above-mentioned research and I/we undertake to ensure compliance with these conditions. Should any departure be contemplated, from the research protocol as approved, I/we undertake to resubmit to the Committee. <u>I agree to submit a yearly progress report</u>. The date for annual re-certification will be one year after the date of convened meeting where the study was initially reviewed. In this case, the study was initially review February and will therefore be due in the month of February each year. Unreported changes to the application may invalidate the clearance given by the HREC (Medical).

Principal Investigator Signature

Date

PLEASE QUOTE THE PROTOCOL NUMBER IN ALL ENQUIRIES

APPENDIX C

From: Hilary Thurling <<u>Hilary.Thurling@wits.ac.za</u>> Date: Tuesday, September 20, 2016 at 4:45 AM To: "Leighton, Kim" <<u>kleighton@devrygroup.com</u>> Subject: RE: Request Permission to your the SCORS instrument.

Dear Kim,

L

Thank you for your reply, and forwarding the SCVORS instrument and guide book to us.

My student is in the process of putting together her proposal for the Post Graduate Assessors Committee.

I am sure she will only be ready for data collection in January/February 2017.

We are hoping to assess the organizational readiness of nursing, pharmacy, physiotherapy, occupational therapy and sports medicine.

I will be in touch and let you know how we are progressing.

Many thanks

Hilary



Hilary Thurling Simulation Coordinator Tel: 011717-2548 Email: hilary.thurling@wits.ac.za Medical School, 4rd Floor Room 4B20



From: Leighton, Kim [mailto:kleighton@devrygroup.com] Sent: 11 September 2016 09:54 PM To: Hilary Thurling Cc: Colette Foisy-Doll Subject: Re: Request Permission to your the SCORS instrument.

Hello Hilary,

So sorry to not have responded sooner-I did get your first email. I've had a lot of international travel this past month and I'm behind!

We are happy to have your student use the SCORS instrument for her Masters work. I'm wondering what your timeline is? The tool is still undergoing data collection as we have had a very low response rate. The data collection is due to end on 10/31/16 and then we will quickly determine whether we have a reliable tool. How does that fit into your student's timeline? Meanwhile, if you'd like to take a closer look at the survey and guidebook, it can be found here: <u>https://sites.google.com/site/scorsfile/</u>

Thanks for your inquiry! Kim

Kim Leighton, PhD, ANEF

Assistant Dean Research & Simulation Faculty Development

DeVry Medical International's Institute for Research & Clinical Strategy 485 US Highway 1 South Building B, Floor 4 Iselin, NJ 08830

c: 402-617-1401 e: <u>KLeighton@devry.edu</u> From: Hilary Thurling <<u>Hilary.Thurling@wits.ac.za</u>> Date: Friday, September 9, 2016 at 3:59 AM To: Kim Leighton <<u>kleighton@devrygroup.com</u>> Subject: Request Permission to your the SCORS instrument.

Dear Kim,

I am a member of the NLN 2015 Simulation leadership programme, and attended your presentation on the development of your SCORS Instrument, at the INACSL conference in Dallas this year.

I am the simulation coordinator for the University of Witwatersrand in Johannesburg, South Africa.

I have a postgraduate MSc student (Nursing) who is interested in assessing organizational readiness of nursing education institutions, for simulation based education.

This will be her research component of her MSc Course work.

I would like to request your permission to use your SCORS instrument with the guideline hand book for her data collection, if possible?

If you are willing for us to use your instrument, could I request that you send us the instrument and access to the guide book plus the correct citation.

We are happy to send you the research proposal and the results of the data collection - if we can assist you in any manner please let us know.

If you require any further information please do not hesitate to contact me at the email address in this correspondence.

With thanks and appreciation

Hilary Thurling



Hilary Thurling Simulation Coordinator Tel: 011 717-2548 Email: hilary.thurling@wits.ac.za Medical School, 4^{ro} Floor Room 4B20



This communication is intended for the addressee only. It is confidential. If you have received this communication in error, please notify us immediately and destroy the original message. You may not copy or disseminate this communication without the permission of the University, Only authorised signatories are thus advised that the content of this message may not be legally binding on the University and recipients are thus advised that the content of this message may not be legally binding on the University and peripersity and recipients are not necessarily the views and opinions of the University and outsiders are subject to South African Law unless the University agrees in writing to the contrary.

This communication is intended for the addressee only. It is confidential. If you have received this communication in error, please notify us immediately and destroy the original message. You may not copy or disseminate this communication without the permission of the University. Only authorised signatories are competent to enter into agreements on behalf of the University and recipients are thus advised that the content of this message may not be legally binding on the University and may contain the personal views and opinions of the author, which are not necessarily the views and opinions of The University and outsiders are subject to South African Law unless the University agrees in writing to the contrary.

This communication is intended for the addressee only. It is confidential. If you have received this communication in error, please notify us immediately and destroy the original message. You may not copy or disseminate this communication without the permission of the University. Only authorised signatories are competent to the permission of the University of University of the without the semission of the University of the without the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the semission of the University of the Without and the contrary.



APPENDIX D

PARTICIPANT'S INFORMATION DOCUMENT AND CONSENT FORM ASSESSING STAFF READINESS FOR SIMULATION IN A HEALTH SCIENCES

INSTITUTION

Dear Respondent,

My name is Miss Tolulope A. Awogbemila, a master's student in the Department of Nursing Education at the University of the Witwatersrand conducting a research study on: "Assessing staff readiness for simulation in a health sciences institution"

This study is designed to examine the readiness of lecturers from the five departments within The School of Therapeutic Sciences to adapting simulation-based education and identify factors preventing or promoting the successful use of the new methodology.

Participation in the study involves completion of a 24-question survey to assist with this investigation which will take approximately 15 minutes to complete. There are no risks or discomforts that are anticipated from your participation in the study. The information obtained has the potential to improve the successful outcomes and efficiency of simulation-based programs.

Confidentiality is assured as names are not required, only the departmental specialty for statistical purposes and information will only be included in the findings.

Participation in this study is voluntary and refusal to participate will involve no penalty. If you have any questions about this project, feel free to contact Miss T.A. Awogbemila, on 0789589221 or Mrs. Hilary Thurling on 0825557003. If you choose to participate, please sign and date the consent form and return it along with the survey to the researcher.

Thank you in advance for the time and effort required to fill out the survey and to assure you that your participation is greatly appreciated.

I have read the above information and hereby freely consent to participate in this study.

Participant's signature

Date

APPENDIX E

Confidential

SIMULATION CULTURE ORGANIZATIONAL READINESS SURVEY

Please complete the survey below.

Thank you!

Tolulope Awogbemila

MSc Nursing student

Dear lecturer,

This survey forms part of my MSc Research titled: "Assessing staff readiness for simulation in a health sciences institution"

This study is designed to examine the readiness of lecturers from the five departments within the School of Therapeutic Sciences to adapting simulation-based education and identify factors preventing or promoting the successful use of the new methodology. The information obtained has the potential to improve the successful outcomes and efficiency of simulation-based programs.

Please respond to the questionnaire below.

DEMOGRAPHIC INFORMATION

- 1) Please select your age group
- 2) Please indicate your gender
- 3) Please select your department

4) Please select your employment status

O 20-30 years 31-40 years
 41-50 years
 51-60 years
 Above 60 years Male
 Female
 Bisexual

- O Exercise Science and Sports Medicine

- Nursing Education
 Occupational Therapy
 Pharmacy and Pharmacology
 Physiotherapy
- O Part-time O Full-time

05-06-2017 13:12

www.projectredcap.org



- Please select the years of experience that applies to you
- 1-5 years
 6-10 years
 11-15 years
 16-20 years
 21-25 years
 Above 25 years

SECTION A: DEFINED NEED AND SUPPORT FOR CHANGE None at All A Little Somewhat Moderately Very Much 0 0 0 0 0 6) To what extent are innovation, experiential learning and quality student experiences clearly described as central to the mission and philosophy of your institution? To what extent has your department clearly defined the 0 0 0 0 0 need to consider simulation--based education (SBE) integration? 0 0 0 0 0 8) To what extent have administrators within your department communicated a clear strategic vision for SBE? To what extent have administrators within your 0 0 0 0 0 department provided a written commitment to SBE? 0 0 0 0 0 10) To what extent have administrators within your department provided funding to support the commitment to SBE? 11) To what extent does your 0 0 0 0 0 department promote the need for SBE based on current evidence, standards, and guidelines? 0 12) To what extent is SBE currently 0 0 0 0 being used as a teaching modality in your department?

13)

05-06-2017 13:12

www.projectredcap.org

REDCap

Con	fidential					
						Page 3 of 6
	To what extent have the educators you work with articulated a need for SBE integration into the curriculum?	0	0	0	0	0
14)	To what extent have the educators in your department verbalized a commitment to SBE integration into the curriculum?	0	0	0	0	0

SECTION B: READINESS FOR CULTURE CHANGE.

To what extent is there a critical mass of professionals who already possess strong SBE:

		None at All	A Little	Somewhat	Moderately	Very Much
15)	Knowledge	0	0	0	0	0
16)	Skills	0	0	0	0	0
17)	Positive Attitudes	0	0	0	0	0

05-06-2017 13:12

www.projectredcap.org



=

Page 4 of 6

	Section B: Readiness for Cul					
		None at All	A Little	Somewhat	Moderately	Very Much
18)	To what extent do administrators support culture change including the efforts required to implement and sustain SBE program integration?	0	0	0	0	0
19)	To what extent are there credentialed or trained simulationists who mentor/coach others, including, other simulationists?	0	0	0	0	0
20)	To what extent does your department have individuals who model SBE best practice?	0	0	0	0	0
21)	To what extent are staff/faculty proficient in the use of technology? (I.e. computer systems, AV and IT systems)	0	0	0	0	0
22)	To what extent are there graduate level prepared researchers available to assist in research to develop new knowledge, appropriate to your department's mission?	0	0	0	0	0
23)	To what extent are librarians available within your institution to help search for evidencebased practice and related simulation resources?	0	0	0	0	0
24)	To what extent are your librarians accessed to search for evidence based practice and related simulation resources?	0	0	0	0	0
25)	To what extent do you believe that now is the right time to implement a culture change to support SBE?	0	0	0	0	0

05-06-2017 13:12

www.projectredcap.org



	SECTION C. TIME, PERSONN	LE AND RESOL				
		None at All	A Little	Somewhat	Moderately	Very Much
26)	To what extent do employees in your department have access to quality technology, including computers, audiovisual equipment, and other institutional technologies?	0	0	0	0	0
27)	To what extent is support available to learn and manage technologies that support education?	0	0	0	0	0

To what extent are fiscal resources available to support SBE in the following areas:

28)	Human resources (simulation personnel)?	None at All	A Little	Somewhat	Moderately O	Very Much
29)	Education?	0	0	0	0	0
30)	Release time to lead integration of SBE?	0	0	0	0	0
31)	Development of physical learning spaces?	0	0	0	0	0
32)	Equipment?	0	0	0	0	0

To what extent are there existing simulation champions (people who will go the extra mile to advance simulation) in the current environment among:

33)	Administrators?	None at All	A Little	Somewhat O	Moderately	Very Much
34)	Clinicians?	0	0	0	0	0
35)	Educators?	0	0	0	0	0
36)	Technology Specialists?	0	0	0	0	0
37)	Administrative Assistants and Support Staff?	0	0	0	0	0



41) Administration?

0

SECTION D: SUSTAINABLE PRACTICES TO EMBED CULTURE None at All A Little Somewhat Moderately Very Much 0 0 0 38) To what extent is the 0 0 measurement and sharing of outcomes part of the culture of the department in which you work? To what extent are decisions regarding SBE influenced by: None at All A Little Somewhat Moderately Very Much 0 0 0 0 0 39) Clinicians? 0 0 0 0 0 40) Educators?

0

0

0

0

www.projectredcap.org



APPENDIX F

	SCORS: Simulation Culture Organizational Readiness Survey								
A.	Defined Need and Support for Change	None at All	A Little	Somewhat	Moderately	Very Much	Scores		
1	To what extent are innovation, experiential learning and quality student experiences clearly described as central to the mission and philosophy of your institution?	1	2	3	4	5			
2	To what extent has your organization clearly defined the need to consider simulationbased education (SBE) integration?	1	2	3	4	5			
3	To what extent have administrators within your organization communicated a clear strategic vision for SBE?	1	2	3	4	5			
4	To what extent have administrators within your organization provided a written commitment to SBE?	1	2	3	4	5			
5	To what extent have administrators within your organization provided funding to support the commitment to SBE?	1	2	3	4	5			
6	To what extent does your organization promote the need for SBE based on current evidence, standards, and guidelines?	1	2	3	4	5			
7	To what extent is SBE currently being used as a teaching modality in your institution?	1	2	3	4	5			
8	To what extent have the educators you work with articulated a need for SBE integration into the curriculum?	1	2	3	4	5			
9	To what extent have the educators in your institution verbalized a commitment to SBE integration into the curriculum?	1	2	3	4	5			
	Subtotal Section A								
в.	Readiness for Culture Change	None at All	A Little	Somewhat	Moderately	Very Much	Scores		
10	To what extent is there a critical mass of professionals who already possess strong SBE:								
а.	Knowledge	1	2	3	4	5			
ь.	Skills	1	2	3	4	5			
с.	Positive Attitudes	1	2	3	4	5			
11	To what extent do administrators support culture change including the efforts required to implement and sustain SBE program integration?	1	2	3	4	5			
12	To what extent are there credentialed or trained simulationists who mentor/coach others, including, other simulationists?	1	2	3	4	5			
13	To what extent does your organization have individuals who model SBE best practice?	1	2	3	4	5			
14	To what extent are staff/faculty proficient in the use of technology? (I.e. computer systems, AV and IT systems)	1	2	3	4	5			
15	To what extent are there graduate level prepared researchers available to assist in research to develop new knowledge, appropriate to your organization's mission?	1	2	3	4	5			
16	To what extent are librarians available within your organization to help search for evidence-based practice and related simulation resources?	1	2	3	4	5			
17	To what extent are your librarians accessed to search for evidence based practice and related simulation resources?	1	2	3	4	5			
18	To what extent do you believe that now is the right time to implement a culture change to support SBE?	1	2	3	4	5			
	Subtotal Section	Potential Score 55							

c	Time, Personnel, and Resource Readiness	None at All	A Little	Somewhat	Moderately	Very Much	Scores
19	To what extent are fiscal resources available to support SBE in the following areas:	w					
а.	Human resources (simulation personnel)?	1	2	3	4	5	
ь.	Education?	1	2	3	4	5	
¢.	Release time to lead integration of SBE?	1	2	3	4	5	
d.	Development of physical learning spaces?	1	2	3	4	5	
e.	Equipment?	1	2	3	4	5	
20	To what extent do employees in your institution have access to quality technology, including computers, audiovisual equipment, and other institutional technologies?	1	2	3	4	5	
21	To what extent is support available to learn and manage technologies that support education?	1	2	3	4	5	
22	To what extent are there existing simulation champions (people who will go the extra mile to advance simulation) in the current environment among:		•/				
а.	Administrators?	1	2	3	4	5	
b .	Clinicians?	1	2	3	4	5	
٤.	Educators?	1	2	3	4	5	
d	Technology Specialists?	1	2	3	4	5	
	Administrative Assistants and Support Staff?	1	2	3	4	5	
	Subtotal Section	c	1	1	1	Potential Score 60	
D.	Sustainability Practices to Embed Culture	None at All	A Little	Somewhat	Moderately	Very Much	Scores
23	To what extent is the measurement and sharing of outcomes part of the culture of the organization in which you work?	1	2	3	4	5	
24	To what extent are decisions regarding SBE influenced by:						
2.	Clinicians?	1	2	3	4	5	
ь.	Educators?	1	2	3	4	5	
с.	Administration?	1	2	3	4	5	
	Subtotal Section	D				Potential Score 20	
	Not Restly: 9-35 A Little: 37-72Somewhat: 73101M Please refer to the SCORS Companion Guidebook for scoring op organization.	olevilely: 109–144– tions before you	-Very Much: 14 u begin scol	5–189 ring your	TOTAL OVI (potential	ERALL SCORE score = 180)	
	SCORS SUMMARY IMPRESSION	Not Ready	Getting Ready	Been Ready But Not Acting	Ready to Start to Act	Past Ready & Into Action Planning	
25	Considering all of the SCORS indicator scores, how would you rate your organization's readiness for SBE integration?	1	2	3	4	5	
26	Looking back 6 months, how would you have rated your organization's readiness for SBE integration?	1	2	3	4	5	
Ple	ot your overall readiness level by marking an "X" on the adjacent	NOT READ	r			A(GREAT DEAL

To cite, please use:

Foisy—Doll, C., & Leighton, K. (2016). SCORS: Simulation culture organizational readiness survey@. An adaptation with permission of the "Organizational Culture & Readiness for System— Wide Integration of Evidence—based Practice Survey@" Fineout—Overholt, E., & Melnyk, B. M. in B. M. Melnyk, & E. Fineout—Overholt. (Eds.). (2015). Evidence—based practice in nursing & healthcore: A guide to best practice. (3rd Ed.). Philadelphia, PA: Lippincott, Williams, & Wilkins.

For permission to use the SCORS tool, please contact either:

Colette Foisy-Dolt foisydc@gmail.com or Kim Leighton: huskerrn@gmail.com